THE FUTURE OF DIGITAL BANKING SERVICES IN MALAWI: EXPLORING THE CHALLENGES AND GROWTH

MASTER OF BUSINESS ADMINISTRATION DISSERTATION

ANNA TINYADE KHUNTHO KAYIRA

UNIVERSITY OF MALAWI
THE POLYTECHNIC



THE FUTURE OF DIGITAL BANKING SERVICES IN MALAWI: EXPLORING THE CHALLENGES AND GROWTH

MASTER OF BUSINESS ADMINISTRATION THESIS

By

ANNA TINYADE KHUNTHO KAYIRA

A Dissertation Submitted to the Department of Management Studies, Faculty of Commerce, in
Partial Fulfilment of the Requirements for the Award of a Degree of Master of Business

Administration

University of Malawi

The Polytechnic

DECLARATIONS

I, Anna Tinyade Khuntho Kayira, declare that this dissertation is my own original work. Where other sources of information have been used, I have acknowledged the information accordingly. As such, this work has not been submitted at any university before in partial fulfilment of any other degree or examination.

Signature:	
Date:	

WORD COUNT: 26,670

CERTIFICATE OF APPROVAL

We, the undersigned, hereby certify that we have read and approve for examination by the University of Malawi, this dissertation entitled *'The future of digital banking services in Malawi: Exploring the opportunities and growth'*

Supervisor:	Dr. Flemmings F. Ngwira, PhD
Signature:	
Date:	
Head of Department: Signature:	Dr. George Dzimbiri, PhD
Date:	
Dean of Postgraduate Studies	s and Research: Ass. Prof. Rhoda Bakuwa, PhD
Signature:	
Date:	

DEDICATION

I dedicate this work to my dear late mother, Mrs. Emily Sunganani Khuntho. She was a great mother and always believed in me. Her beautiful spirit will always live on in us.

ACKNOWLEDGEMENTS

First and foremost, I would like to thank the Almighty God for the gift of life and good health as I was pursuing this programme at the University of Malawi. If it were not for God, this milestone would not have been realised. All praises and honour to the Almighty God.

I would also like to thank the management, staff and customers of the following banks for granting me an opportunity to interview them for this dissertation to materialise; FDH Bank, Ecobank Malawi, National Bank of Malawi plc, NBS Bank and Standard Bank Limited.

I would like to thank Management of the National Bank of Malawi plc, for granting me an opportunity to leave work on Fridays and Saturdays to pursue this dream. I would also like to thank my colleagues who covered me up in those days when my presence was needed at the office but they sacrificed to make my dream come true. A special mention should go to Mrs. Ivy Mbwelera Mshali, the then service centre manager for NBM Chichiri for being so tolerant and thoughtful throughout my study of this programme, may she do the same to others.

To my great supervisor, Dr. Flemmings Fishani Ngwira, I would like to thank him for his outstanding coaching skills, and for being a great leader and supervisor; I would not have made it this far without his great mentorship. And to all the lecturers that moulded me to be where I am today, may the good Lord bless them all.

Last but not least, great appreciation should go to my dear husband, the wind beneath my wings, Wyson Kayira, for being a great companion throughout my study of this programme. I would like to thank him for taking care of our girls while I was away from home for most of the weekends. It was not easy, but he made everything look easy because he wanted to see me excel. May the good Lord reward him in ten-fold. And to our daughters Miranda and Yolanda, I want to thank them for always being wonderful kids while mum was away for most of the weekends, they gave me a reason to push on. And to my dear father, my sisters Chikondi and Linda and my brother Yamikani, I want to thank them for all the support that they rendered to me during the period of study.

ABSTRACT

Globally, banks continue to invest in Information and Communication Technology (ICT) to accelerate the growth of digital banking. Banks have now migrated from the usual "brick and mortar" to what is called "click and mortal". Developing countries have not been left out, they are also doing anything possible to compete in the global market. This research aimed to explore the knowledge and usage of e-banking services, identify challenges that affect the growth of digital banking in Malawi and explore the growth strategies for the acceleration of digital banking services. Five banks in the Southern Region took part in this research where four customers from each bank were interviewed. Key informants from the banks were also interviewed to understand the challenges and possible remedies banks would use to accelerate digital banking growth. Using thematic analysis to analyze the data, the study came up with the following themes: lack of awareness (of banks' products and services), banks' complacence, the need for change and external factors challenges. Participants highlighted the need for the banks to act with urgency whenever they present their grievances to the bank. Automation of more services on e-banking was also highlighted as one of the areas that the banks need to improve on to keep away customers from the banking halls. The desire for cyber security enhancement and sentitisation of the banks' products are services were also emphasized. These results have significant implications for policymakers, who have an important role to play in making sure that the country's e-banking infrastructure and services are improved to have a conducive environment for digital banking adoption. The research was cross sectional as five banks were sampled for this research. It is therefore recommended that for future research, all the banks in the country should take part in this research to have a better perspective of the delivery of digital financial services. It is also important that for further research, third parties in the service delivery of digital financial services be included to also have a better understanding from all concerned parties.

TABLE OF CONTENTS

DECLARATION	ii
CERTIFICATE OF APPROVAL	iii
DEDICATION	iv
ACKNOWLEDGEMENTS	v
ABSTRACT	vi
LIST OF TABLES	X
ABBREVIATIONS AND ACRONYMS	xi
CHAPTER ONE: INTRODUCTION	1
1.1 Background of the study	1
1.2 Problem statement	3
1.3 Main objective	5
1.4 Specific objective	5
1.5 Research questions	5
1.6 Rationale of the study	5
1.7 Definition of key terms	7
1.8 Structure of the thesis	7
CHAPTER TWO: LITERATURE REVIEW	9
2.1 Chapter overview	9
2.2 Theoretical framework	9
2.3 Technology acceptance theory	9
2.4 Innovation diffusion theory	10
2.5 Digitisation of the banking services industry	11
2.6 Digital banking services from the banker's perspective	14
2.7 Digital banking services from the customer's perspective	17
2.8 Chapter summary	20

CHAPTER THREE: RESEARCH METHODOLOGY	21
3.1 Chapter overview	21
3.2 Research Philosophy	21
3.3 Research design	21
3.4 Population	22
3.4.1. Study setting	22
3.4.2 Characteristics of the sample	22
3.4.3 Sampling technique	22
3.4.4 Sample size	23
3.5 Data collection	23
3.6 Data analysis	24
3.7 Ethical consideration	25
3.8 Validity and reliability	25
3.9 Chapter summary	26
CHAPTER FOUR: RESULTS AND DISCUSSION	27
4.1 Chapter overview	27
4.2 Customer's knowledge and user experience of digital banking service	31
4.2.1 Lack of awareness	31
4.3 Challenges hindering growth of digital financial services in Malawi	33
4.3.1 Banks' complacence	33
4.3.2 External Factor Challenges	36
4.4 Banker's and customer's perceptions of the growth of digital banking services	38
4.4.1 The need to embrace change	38
4.5 Discussion of results	41
4.5.1 Customer knowledge of e-banking services	41

4.5.2 Challenges to E-banking growth strategies	43
4.5.3 Perceptions on growth of digital banking services	46
4.4 Chapter summary	50
CHAPTER FIVE: SUMMARY OF FINDINGS, CONCLUSIONS AND	
RECOMMENDATIONS	52
5.1 Conclusion	52
5.2 Recommendations	53
5.3 Limitations	54
5.4 Areas for Further research	54
REFERENCES	55

LIST OF TABLES

Table 1: Demographic information of the sample for the qualitative research	27
Table 2: Themes and sub-themes	29

ABBREVIATIONS AND ACRONYMS

ATM Auto Teller Machines

ICT Information, Communication & Technology

POS Point of Sale

USSD Unstructured Supplementary Service Data

MNO Mobile Network Operator

E-banking Electronic Banking

CHAPTER ONE: INTRODUCTION AND BACKGROUND

1.1 Background of the study

The advancement in Information and Communication Technology globally has changed the way firms conduct their businesses. The shift of digital technology from mechanical and analogue electronic technology started as early as the 1940s and this led to the adoption of digital computers and digital record keeping (International Monetary Fund[IMF], 2021). The Internet was developed in the 1960s and 1970s and became one of the prominent applications of digital technology (IMF, 2021). The rapid increase of digital technology has led banks to be one of the leaders in adopting these new technologies (Borges et al., 2020). The banking industry, through the introduction of digital banking, has responded to the new needs and changes in the lifestyle of its customers (Shcherbatykh et al., 2021).

The banking industry has witnessed how the "brick and mortar" type of business, where people traditionally used to visit the banking halls to transact, to know what is called the "click and mortar" where most of the banking transactions are being done at the click of a button (King, 2018). The younger generation expects faster services more connected to their smartphones, is more attracted to international travel, and is aligned with their mobile and convenient lifestyle (IMF, 2021). The banking industry therefore comes in to answer to the modern way of living that the younger generation wants to adapt to. A study has shown that banks appear to have been the digital leaders, likely benefiting from a sound digital infrastructure, a strong legal and business environment, and healthy competition (IMF, 2021). As such, it is inevitable for the banks to follow the technological advancements that are being experienced globally.

The COVID-19 pandemic has also necessitated the need for banks to change how they conduct their business (Reserve Bank of Malawi, 2020). It has provided a great boost to emerging digital banking services as it provides more convenient access to financial services (Borges et al., 2020). During the peak of the pandemic, many people were afraid to go out of their homes for fear of catching the virus (Andrade et al., 2022). Others were also limited to going out because they were either in isolation or they were looking after their sick relations. Life still had to go on, it was, therefore, electronic banking that was more convenient for people to keep in touch with the business world (Reserve Bank of Malawi Monthly Economic Review, 2022). This made life easier for most of the people who were unable to go out of their homes to do traditional banking.

There are different ways in which banks can apply digital transformation to their services. Firstly, using websites is the customer's first touch point with the bank (Borges et al., 2020). A customer can know the products and services that a bank offers before visiting the bank. This helps customers make informed decisions on what to do at the bank. The use of mobile applications is also another way that banks have applied digital transformation. Mobile applications are where the banks have applications that people can download to access their bank accounts through their phones (Kaur et al., 2021). The use of Automated Teller Machine (ATM), internet banking, and Point Of Sales (POS) machines have all made it possible for banks to be technologically advanced therefore keeping in touch with their customer base (Skinner, 2014).

As much as there are these technological advancements that banks must adopt to stay competitive in the market, there are also challenges that are hindering the advancement of digital banking on the global level. It has been noted that about 49% of American adults do not participate in online banking (Revathi, 2019). This is the case because the older generation prefers traditional banking to modern banking. Security is also one of the drawbacks of digital banking. In 2015 in the United Kingdom, roughly 130 million British pounds were stolen from online bank accounts through fraud (Revathi, 2019). Frauds contribute to people's continued preference for traditional banking channels, which are considered safer for customers (Rabiu et al., 2019a). Additionally, the complexity of transactions poses a challenge to the global progress of digital banking. It has been observed that depositing and withdrawing money from an online bank is significantly more difficult and time-consuming (Revathi, 2019). Technical issues have also been attributed to be one of the major challenges in the advancement of digital banking worldwide (Selvam et al., 2018). Despite the heavy investments in digital technology, banks still find themselves dealing with the challenges to do with technology.

There are also opportunities in the digital banking world that banks can explore and benefit from. The e-banking has brought in an increase in the volume of deposits from customers (as cited by Agwu & Carter, 2014; Taiwo & Agwu, 2017 & Severino et al., 2015). The deposit mobilisation is important for the banks as these funds can be used to lend to customers which can in turn lead to more interest for the bank and the deposits also help with the bank's cash flow (Banke & Yitayaw, 2022). Digital banking is generally viewed as a means of providing traditional services in a modern way as it aims at providing speedier and time efficient transactions for customers with little or no paperwork involved (as cited by Hossain et al, 2013 in Taiwo & Agwu, 2017). This is an opportunity for both the customer and the bank as the customer can transact at the tip of a button

which is convenient and fast, and it is also important to the latter as it saves from the establishment of more physical banks which are generally costly than managing a virtual bank (IMF, 2021). Digital banking is largely motivated by the prospects of minimising operating costs while maximising operating revenue (Selvam et al., 2018).

It should be noted that digitalisation has changed the face of banks. It was noted that globally, in 2019, digitalisation cut the operating costs of banks from 50% to 30% due to fewer branches and employees (as cited by Citi, 2019 in IMF, 2021). The initial investment of new technology is usually costly as the bank must invest in internal and external hardware, new software, and new staff with information technology expertise, however, the recurrent costs in the medium to long term are expected to be lower as banks now spend less on system maintenance (IMF, 2021). Digital banking has also enhanced operational efficiency and it has allowed for financial inclusion as mobile wallets have made it possible for those without accounts to access financial services just like those who have bank accounts (Rabiu et al., 2019b).

Malawi, being a developing country economically, has not been spared by these digital technological advancements. As a country, it has experienced several digital advancements such as the use of Internet banking, mobile banking, the use of Automated Teller Machines (ATM), mobile wallets, and debit and credit cards which have necessitated a smooth flow of business (Rabiu et al., 2019b). Banks have taken advantage of an increase of people using smart gadgets such as smartphones and tablets to tap into this market (Shcherbatykh et al., 2021). Despite these digital advancements, the banking industry has seen several challenges such as high internet prices, low electricity coverage, lack of digital skills, fraud, and traditional hindrances, among other challenges (MACRA, 2020), that must be addressed to find solutions that will mitigate the above challenges. It can be noted from the above literature that despite the challenges that banks globally are facing, there are more opportunities that Malawian banks can embrace to stay competitive. It was therefore the aim of this research to explore those challenges that were hindering the growth of digital banking in the country and explore the growth strategies that would help the acceleration of digital banking in the country.

1.2 Problem statement

The development of digital banking in Malawi dates back to 2002 when the Reserve Bank of Malawi (RBM) established the Malawi Switch (then Malswitch now National Integrated Technologies Limited [Nitl]) to facilitate the implementation of a clear, secure, and guaranteed e-

payment infrastructure (Majanga, 2016). Since the establishment of Malswitch, the penetration of digital banking has been at a snail's pace. People still rely on traditional banks to transact, unlike the digital bank which is faster and more convenient to the customers (Taiwo & Agwu, 2017). Banks have invested heavily in technology as can be witnessed with frequent system upgrades by the service providers. As much as there are these investments which show that there are opportunities in digital banking, the banking industry is still facing more challenges both from the customers' perspective and the banks' perspectives in terms of fraud, system downtime, high internet prices, and low deposit volumes. The much-invested digital technology is not yielding what it ought to yield.

As the long queues at both the ATMs and banking halls have now become the order of the day, this has resulted in bad publicity for the banks. People have now lost trust in the banks that they are unable to solve in-house problems which have resulted in long queues. The system downtime has also affected people's daily lives as they are not able to transact at their convenient times. In Malawi, there is a high number of the unbanked (Malawi Population and Housing Census, 2018) which the banks can take advantage of, but with the challenges that the banks are facing such as fraud, and intermittent system challenges, it is hard to convince such people to open accounts with banks which they can benefit from. The technological challenges have also affected the banks' revenue since most of the digital transactions are chargeable (Revathi, 2019), but where people are not making use of these channels, then the banks are losing out on revenue. Since more people are still resorting to traditional banking than electronic banking, banks are spending more on the establishment of more physical branches and have more manpower to serve such customers which is very expensive for the banks (Selvam et al., 2018). If the above challenges were resolved, the banking industry would see a boost in their deposits, revenue as well as operational efficiency which would all lead to more revenue.

With the above-highlighted issues, it was important that the problems that the banking industry was facing would be looked at seriously and understand from both the customers' and banks' perspectives why such problems were still there. From the bankers, it was also important to seek from them the opportunities and growth strategies they visualise for the future of digital banking in the country if the challenges highlighted above were to be resolved. It was therefore the purpose of this research to identify the challenges that were being faced by banks in terms of the acceleration of digital banking in the country and explore opportunities in digital banking for a desired future in the digital banking industry that people would like to witness.

1.3 Main objective

The main objective of this study was to explore the knowledge and usage of e-banking services and identify challenges that affect the growth of digital banking in Malawi.

1.4 Specific objective

The following were specific objectives for this study;

- a. To assess customer's knowledge and user experience of digital banking services
- b. To establish the challenges that are hindering the growth of digital financial services in Malawi.
- c. To explore the bankers' and customers' perceptions of the growth of digital banking services

1.5 Research questions

- a. What is the customers' knowledge and user experience of digital banking services?
- b. What are the challenges that are hindering the growth of digital banking services in Malawi?
- c. What are the bankers' and customers' perceptions on the growth of digital banking services?

1.6 Rationale of the study

Malawi as a developing country, has set goals that it wants to achieve by the year 2063 ("Malawi Vision 2063 Paper," 2020). The Malawi Vision 2063, in terms of Information and Communication Technology, states that "the country shall have a robust ICT infrastructure with cross-country coverage of reliable and affordable services fostering technological adoption and digital access" ("Malawi Vision 2063 Paper," 2020). As is the case now, the country has not achieved a robust infrastructure in terms of ICT as some areas in the country still face network challenges that people are unable to inclusively take part in digital banking services (Malawi Population and Housing Census, 2018). As it is the vision of this country to see to it that the country by 2063 it has advanced in terms of ICT, this research would assist in fostering this vision by compiling the challenges that the banking sector is facing in terms of the adoption of digital banking services and opportunities that would help in the implementation of Malawi's Vision 2063.

Secondly, the Reserve Bank of Malawi is advocating for a cashless society where people need to embrace the use of plastic money when making payments (Reserve Bank of Malawi Monthly Economic Review, 2022). This has a favourable consequence in the sense that it would improve

financial inclusion (IMF, 2021). As of 2021, out of a total population of 19.3 million in the country, only 23% had access to banking activities leaving out a huge figure of 77% of people not involved in any banking activities (*Malawi Key Country Statistics*, n.d.). This is worrisome as being in the 21st Century, as a country, there should have been strides made in terms of digital advancements. This research would help to find the challenges that people face that make them not take an active part in financial inclusion activities. The research would also help to find the opportunities that the banks can make to include more people in digital financial activities.

Lastly, as digital technology has advanced, banks have often been leaders in adopting these new technologies (IMF, 2021). Financial institutions are heavily investing in e-banking infrastructure to improve service delivery to their customers (Borges et al., 2020). As much as this is the case, banks still face challenges in the adoption of digital services such as low electricity coverage, high internet prices, high cost of smart devices, lack of digital skills, cultural issues and other factors that are yet to be explored (National Survey on ICT Access and Usage, 2019). It was therefore important that these challenges that people were facing needed to be looked at critically and explore strategies that would help banks reap the heavy investments that they are making in digital financial services.

Malawi as a developing country, the digital advancement is not yet at its peak as is the case with developed countries. For digital maturity to reach in the country, there is a need for more research to be conducted to understand the digital banking platforms and developments that are happening globally. A review of the literature has shown that there are not many studies that have been conducted in terms of digital banking in the country. A few of the studies that have been conducted on digital banking have given much attention to ATMs and mobile banking as these were the first platforms that were first launched on the market. Digital banking (which encompasses Internet banking, mobile banking, ATMs, POS, and debit cards) has not been given much attention just like the two platforms have been given the attention. The challenges that were explored in other surveys, may not be the same challenges that the banking system is facing today. Therefore, this study sought to understand the challenges that customers face today in the adoption of digital banking.

With an increase in the number of banks and customers as well, there are opportunities that probably banks are missing out on that can accelerate the adoption of digital banking in the country. This research was important therefore to explore those opportunities that would help banks to drive

their business and in the long run, accelerate the economy of the country. If the banking industry is to improve, then there is a need for researchers to well understand the volatile environment that the banking industry is operating in. Therefore, as banking transactions are the order of the day, the results from this research would help to understand the ever-changing environment in which banks are operating in. The results from this study would also help to contribute to the academic literature on digital banking in Malawi as the literature currently is limited and help those wishing to undertake similar studies to have more access to literature on which they can base their studies. Even though qualitative results cannot be statistically generalised, they can be generalised or transferred to other contexts or settings. The results from this research can therefore be generalised in other settings apart from the setting that was used to conduct this research.

1.7 Definition of key terms

The key definitions the study used are as follows:

Digital banking

An electronic payment system that allows customers of financial institutions to conduct financial transactions through the financial institution's website or through the use of a smart gadget.

Mobile banking

An act of making financial transactions on a mobile device (cellphone, tablet, etc.).

Auto Teller Machines

An electronic banking outlet that allows customers to complete basic instructions without the aid of a bank official.

1.8 Structure of the thesis

This chapter has given a background of where digital banking originated and how developing countries have adopted e-banking. The chapter also introduced the problem statement, main and specific objectives of this study and rationale for this study.

Chapter two presents the literature review of this study. This chapter reviewed the literature about the studies that have been done in developed countries and how developing countries are adopting digital financial banking. The chapter also introduces the theories that were used to support this research.

Chapter three introduces the methodology that was employed to gather data that was useful for this research.

Chapter four discusses the results that were found from this research using Nvivo 12 plus where thematic analysis was used to come up with codes and later themes that were dominant in this research.

Chapter five summarises what has been discussed in the previous chapters and recommends areas for further study and limitations for this study.

CHAPTER TWO: LITERATURE REVIEW

2.1 Chapter overview

The chapter discusses the theories that inform this study. The Technology Acceptance Theory (TAM) and the Diffusion of Innovation underpinned this study. The study then reviews the literature on digital banking adoption both in developed and developing countries. The discussion establishes a gap in how digital banking is being implemented in developing countries, and this study aims to fill this gap.

2.2 Theoretical framework

This research was guided by the Technology Acceptance Model (TAM) and Innovation Diffusion Theory (IDT). The two theories are explained below in connection with the issue that was at hand. Technology Acceptance Theory is the anchoring theory for this research as it explains how technology is accepted and how the perceived usefulness of a technology advocates for people to accept a technology.

2.3 Technology acceptance theory

The Technology Acceptance Model theory was developed by Davis in 1989 to examine the acceptance and usage of technologies (Chindudzi et al., 2020). The concept is centred on analysing the user's adoption behaviour concerning the internal and external factors of technology. The theory is centered on two main perspectives which are perceived ease of use and perceived usefulness which are the two main factors that influence electronic banking adoption. The perceived ease of usefulness is defined as the degree to which a person believes that using a specific technology will help them to perform better. In contrast, the perceived ease of use is the degree to which a person can treat and interact with technology without complications or effort (Aburbeian et al., 2022). This therefore illustrates that, if the technology is easy to use, then this will improve its usefulness. If a user interacts with a new technology that is useful, easy to use and valuable, then the user will have a positive attitude which will increase the intention to use that technology (Aburbeian et al., 2022). This theory has an impact on the attitudes towards technology use and as a result, it will influence behavioral intention.

One of the issues that were highlighted in the literature was that some customers found it hard to access online banking because of the processes involved and customers found it difficult to do so. The literature also revealed that about 49% of adults in America do not take any part in online

9

banking as they prefer the old traditional way of banking. Customers are being hindered from using online banking because they have an attitude that online banking is for the elite, and it is hard to operate. If the banks can take a step further by educating their customers on how they can go about the use of digital banking, customers would then change their attitudes and they will see fit to take part in such practices. If this is repeatedly told to the customers, the ease of use of online banking would be easy which would lead to the usefulness of such innovations.

2.4 Innovation diffusion theory

The Innovation Diffusion Theory (IDT) was developed by E.M. Rogers, a communication theorist in 1962. This theory attempts to explain and describe the mechanisms of how new inventions (in this case, online banking) are adopted and become useful (Chindudzi et al., 2020). Not all inventions are adopted, even if they are good, it may take a long time for that innovation to be adopted and used. The resistance to change is one of the factors that may hinder the diffusion of change of a certain innovation. Cultural beliefs, attitudes, and behaviours are some factors that may advocate for resistance to a technological change. The rate of adoption of innovations will also depend on how an organisation perceives its relative advantage, compatibility, triability, observability and complexity (Chindudzi et al., 2020). If a firm perceives such an innovation to be useful, then it will have all it takes to make sure that innovation is more visible to people who will later adopt that technology. The adoption of such innovations is quicker in banks that have strong internet access and robust IT departments than in banks without (Rogers, 2005 cited in Kombe & Wafula, 2015). It is therefore important for banks to have strong IT departments that well know what they ought to do and how to do it for innovations to be diffused to their customers.

Not all inventions are adopted, even if they are good, it may take time for such innovations to be accepted and used by the customers. Some of the factors that hinder the diffusion of innovation are the beliefs, attitudes, cultural practices and behaviours that people are engaged in. The rate of adoption will also depend on how an organisation perceives its relative advantage, compatibility, and complexity. If an innovation tends not to be not very complex, a firm will try as much as possible to speed the rate of diffusion, but if a firm tends to see an innovation to be very complex, then the rate of diffusion will also be slow as the organisation does not pay much attention to that innovation. This entails that there are innovations that are being advocated for by the banks, but their rate of diffusion is low depending on how the banks handle such innovations. It is therefore the responsibility of banks to make sure that when innovations have been developed, the diffusion

rate should be high by marketing that product and making their customers aware of such innovations.

2.5 Digitisation of the banking services industry

Digital banking is a system which enables banking transactions like payments, deposits, and withdrawals of cash virtually with the help of the Internet rather than visiting the physical bank (Sharma & Dubey, 2022). The terms internet banking, online banking, and electronic banking (e-banking) are used interchangeably to mean digital banking (Sharma & Dubey, 2022). Digital banking has changed the face of banking globally. The banking industry has changed from the usual "brink and mortal" to what is now called "click and mortar" since most of the banking transactions are done at the click of a button (King, 2018). To stay competitive in the market, banks have taken a central and leading role in terms of the advancement of digital banking. Since most of the business transactions require the involvement of banks, this has necessitated banks to build a strong brand for themselves to be more visible.

The shift from mechanical and analogue electronic technology to digital technology started as early as the 1940s and this led to the adoption of digital computers and digital record-keeping (IMF, 2021). In the 1970s, home computers were introduced but they were not widely used until the 1990s. The Internet was developed in the 1960s and 1970s and it became one of the prominent applications of digital technology (IMF, 2021). In late 2005, the use of the internet had reached a population of about one billion people (Digitalisation and the Future of Work in the Financial Services Sector, 2021). The rapid rise of technology is one of the key developments that have led to the diffusion of digital technology.

As digital technology has advanced, banks have also been the leaders in adopting these new technologies. Citibank in the United Kingdom was the first bank to install an ATM in 1977 but customers only started using it regularly in the 1980's (IMF, 2021). Online banking was piloted in the 1980s by Chemical Bank but it was not widely used until the 2000s with more widespread Internet usage (Selvam et al., 2018). With computer and internet proliferation, a younger generation of customers has quickly adapted to the newer technology and they are demanding higher quality and more digital-based services (IMF, 2021). The new technology has also allowed banks to boost their operational efficiency with fewer operational staff and bank branches.

There are different ways and platforms on which customers can have access to digital banking. Firstly, it is the use of Automated Teller Machines (ATM). An ATM is a computing device that

allows users of a financial organisation to perform financial transactions automatically with the use of a card (Rahman & Rani Saha, 2018). Using an ATM, customers can access their bank accounts at any time of the day even when banks are closed. Customers can pay their utility bills, make cash deposits and withdrawals, transfer funds between accounts, enquire about their balance, mobile tops, mini statement enquiry and cheque book requests (Rahman & Rani Saha, 2018). The coming in of ATMs has helped with the financial inclusion drive where ATMs are placed in remote areas where there are no physical branches and people have the chance to access their bank accounts (Dasho et al., 2016).

Mobile banking has also become one of the widely used digital banking modes. Mobile banking is referred to as interacting with bank facilities and services through smartphones to electronically complete financial and non-financial transactions without the need to directly interact with bank employees (Payne et al., 2018). Mobile banking has become one of the most commonly used modes of digital banking. To the banks, it creates operating efficiencies, enhances productivity, and provides payroll cost savings, and to the customer, it provides satisfaction, time management and decreased need to interact with the bank employees (Payne et al., 2018). Unlike ATMs where a customer only makes a transaction with the physical presence of an auto teller machine, with mobile banking a customer can initiate a transaction if there is internet availability or access to a network. It is because of this reason that the younger generation is more obsessed with this application as they demand convenient and better customer service (IMF, 2021).

Internet banking is also fast becoming one of the commonly used methods of digital banking. It is a method of banking which allows users to conduct financial transactions via the Internet (Kaur et al., 2021). Internet banking, also known as web banking, offers most of the services that are normally found at the traditional bank. One can send funds to another account, request a cheque book, pay utility bills, check for balance, and apply for a loan using Internet banking. Internet banking is now becoming one of the commonly used methods of digital banking because customers can access it at any time of the day worldwide. Internet banking has made it more convenient for travellers who usually need convenience when they travel to make payments using their local accounts. As much as Internet banking has benefits such as these, most people do not patronise it just like mobile banking. It is therefore an opportunity for banks to advertise this product which can coax more customers and at the same time, banks will be able to collect more revenue from the use of this product.

As we are living in the digital era, no one can run away from the use of digital banking. In one way or the other, we have been affected by the activities of digital banking knowingly or unknowingly. As digital banking has come in to stay with the advancement in technology globally, some advantages are associated with the use of digital banking to both the banks and users. E-banking has proven to be time-saving for users (Devulapalli & Oruganti, 2021). Customers do not need to travel long distances to transact at the traditional bank. Banking transactions can be done in the comfort of one's home. Instead of queuing at the physical bank, which is time-consuming, customers can use that time to do other equally important things.

Secondly, the availability of digital banking at any time of the day has also provided flexibility and convenience to customers (Devulapalli & Oruganti, 2021). The physical banks provide a barrier in that, once banks have closed, customers do not have any access to the bank even if they have an emergency. But with the coming of digital banking, customers can withdraw money, make transfers, or pay for utilities in the comfort of their own time. This has also provided convenience to the customers as they do not have to worry about travelling to the bank, but rather, they can transact at any time of the day. Banks have also benefitted in that, instead of extending more banking hours to allow their customers to transact at the bank, which is costly as the bank has to foot bills that come with the extension, such as, paying staff allowances and overtime, providing transportation to their staff, online banking has helped to cut such costs. Employees have also benefited in that, banking halls close at good time which helps them to get refreshed in readiness for the next day's work and productivity is also enhanced for the satisfaction of customers.

A recent study by Accenture (2019) suggested that digital maturity is associated with increased profitability, that is, digital advanced banks have on average experienced an overall increase of Return On Equity (ROE) of 0.9% between 2011 and 2017 (IMF, 2021). The study also showed that the adoption of new technology has large profitability in the long run although the initial investment is usually costly as the bank needs to invest in internal and external hardware, new software, and new staff that have information technology expertise fit for the job (IMF, 2021). The recurrent costs in the medium to long term are expected to be lower as the bank spends less on inefficient systems. Digitalisation could also cut the bank's operation costs by 30% to 50% since there will be less need for more branches and employees will not have to be many to serve customers (Citi, 2019). Operational efficiency will be enhanced and at the same time, banks will save on operational costs.

To substantiate the point above which says that digital banking is associated with increased return on equity, a recent study by Dickson and Priyar, 2023 (Dickson & Priyar, 2023) on NBS Bank plc (a Malawian bank), showed that the bank has been positively reporting ROE and Return On Assets (ROA) since it embarked on the digital transformation drive. The profitability has been favourable to the bank ever since the bank embarked on digital transformation. The bank has also grown in terms of deposit mobilisation, assets, and customer base since it embarked on the digital transformation drive. This shows that with digital banking, there is hope that banks will do well once the required resources are put together and this will lead to the growth of banks and at the same time improve operational efficiency and growth of banks in all dimensions.

2.6 Digital banking services from the banker's perspective

As technological advancements are taking centre stage, banks have a role they need to play to drive the initiatives that are being accelerated to have conducive digital banking in the country. This section discusses both the opportunities and challenges that are being faced by both developed and developing countries, including Malawi, in advancing digital banking.

To begin with, since the inception of digital banking, the banking industry has been faced with fraud perpetrated by both employees and outsiders (Bunker, 2020). It is worrisome to note that instead of employees being the custodians of private information about their systems, they are the ones who let loose such information to outsiders who later defraud the banks. It was noted by Bunker (2020) that half of the security incidents were perpetrated by bank employees. Apart from the employees being the ones who perpetrate fraudulent transactions, outsiders have also advanced in their tactics on how to defraud banks. The fraudsters are usually ahead of the banking systems and try to make whatever effort they can to stay ahead of whatever they do. In the United Kingdom (UK) in 2015, roughly 130 million British Pounds were stolen from online bank accounts through fraud (Revathi, 2019). For an advanced country like the UK to be defrauded all such huge sums of money is worrisome, let alone for developing countries that are not yet at the peak of technological advancements.

Technical challenges have also been noted to be one of the challenges in the advancement of digital technology (Revathi, 2019). Developing countries lack the expertise to advance in digital technology; the banks lack experienced personnel that can handle the new technology that is supposed to be used daily. This has therefore resulted in system downtime most of the time which gives a bad picture for business (Dickson & Periyar, 2023). Developing countries are also having

challenges in buying modern software and hardware that is supposed to be used to match international standards (Devulapalli & Oruganti, 2021). It should be noted that the equipment that is supposed to be used is quite expensive, as such, most banks prefer to use the same old obsolete equipment that cannot manage to take care of the overload of transactions that happen daily. This has therefore let down customers who still resort to using the old traditional way of banking which does not inconvenience them, unlike digital banking which has hiccups now and again.

Low broadband internet penetration is also one of the factors that are impeding the advancement of digital technology in developing countries including Malawi (Devulapalli & Oruganti, 2021). Unlike developed countries that are now using 5G internet, most mobile operators in the country still use 4G except for Telekom Networks Malawi (TNM) which has recently announced the launch of 5G network. The country is yet to witness how TNM's internet will be performing, but still, the banking sector cannot rely on TNM to provide the much-wanted internet. Internet access is also expensive in the country unlike in developed countries. Users fail to subscribe to the internet because of the exorbitant charges (Malawi Population and Housing Census, 2018). As such, people still resort to traditional banking which is cheaper for them unlike buying data bundles. The number of people who have internet access is still low, which stood at 17.8% against a population of 19.39 million people in 2018 (Malawi Population and Housing Census, 2018). This just shows how challenging it is for the banking industry to penetrate through, but at the same time, there is hope for financial inclusion since about 80% of the population does not have access to the internet. This gives room for banks to explore more and find ways on how they can take these people on board.

Old traditional habits are also a major challenge that banks must overcome to advance in digital banking technology (Revathi, 2019). People are so fixated on the old way of banking instead of exploring the new way of banking which is convenient and fast. In America, about 49% of adults do not participate in online banking, they still prefer traditional banks to digital banking (Revathi, 2019). People prefer to physically meet the bank staff and transact unlike using smartphones and laptops which do not have the human touch. It has been discovered that people love the human interactions at the bank as they can enquire for whichever transaction, they want to know, unlike the gadgets which lack the information that people may need. This is therefore a drawback to the banks as the banks still must increase their capacity in terms of opening more branches and increasing more manpower to serve those customers which is costly in the long run to maintain.

With the introduction of mobile wallets, banks also must face competition from mobile network service providers for customers. Mobile wallets have now taken centre stage in terms of usage. In Malawi, the number of mobile wallet users stood at 12.2 million in December 2022 from 10.7 million in December 2021 representing a 14.1% increase (Reserve Bank Of Malawi, 2022). The active of active digital banking subscribers stood at 1,500,000 in the same period last year. This shows that mobile wallets have taken the centre and most people prefer to use the wallets probably because of their easy registration, accessibility of any other phone and the mobile wallet agents that are easily found in most parts of the country. The mobile network is also usually stable, unlike digital banking which usually relies on the accessibility of the Internet. This has made most people prefer the wallets which have paused a challenge in the acceleration of digital banking in the country.

Despite these challenges, there are also opportunities that banks can explore that can help them accelerate their business and in the long run improve operational efficiencies. This section will discuss the opportunities from the literature that banks can explore.

The Reserve Bank of Malawi issued a directive in 2022 that all merchants that have a turnover of MWK10,000,000 per month should have at least a POS machine at their disposal while those that are below this threshold should have at least one digital banking payment that customers can use (Reserve Bank Of Malawi National Payments Report, 2022). This directive has come in place to encourage people to run away from the use of cash and instead make use of these digital platforms which are convenient and safe. This directive also encourages people to use plastic money thereby advocating for a cashless society just like other banks are doing in the developed countries. This is an opportunity for the banks to boost, improve and market their digital platforms for people to get familiarised with them. If the banks are to take these initiatives seriously, then they can see a growth in the number of accounts being opened, and growth in digital utilisation but also this will result in volumes of deposits being realised. If the volume of deposits can be increased, these funds can be used to lend to their customers which in turn the banks can realise more interest.

It has also been noted that one of the challenges that banks are facing in the drive towards bank digitisation is the culture and preference for the traditional way of banking. If the banks can work well with the RBM, as the regulator of financial entities in the country, and come up with literacy programmes that should support people's attitudes towards digital banking, then this would help banks to improve the numbers of people that are on digital platforms (Reserve Bank of Malawi

Monthly Economic Review, 2022). There are different myths and beliefs that people have concerning digital banking. If these beliefs and attitudes can be corrected, then the banking industry can have a boost in terms of users. It has also been observed that most of the customers do not know how to go about for them to be on digital platforms. If banks can work out these shortfalls, this will pause as an opportunity to banks on how best they take on board most of the customers who do not have access to digital banking.

In India, the government has taken the initiative to introduce internet banking into the school curriculum (Devulapalli & Oruganti, 2021). India is one of the countries that is well advanced in terms of technological advancements. By introducing digital banking into the curriculum, they have seen that this is one way of entrenching the upcoming generation to be well-versed with the digital advancements that are happening globally. If other countries can also emulate the same just as India is doing, this would help children to know what is involved in the digital world. This younger generation will not have a problem later when they are older because they will have been birthed about the pros and cons of using digital advanced gadgets.

Digital banking can also be used as a competitive advantage for banks to excel and do better than other banks in the industry (Devulapalli & Oruganti, 2021). As much as other challenges in digital banking such as low broadband penetration are beyond the bank's ability, banks can use some of the challenges that they face to be their competitive advantage. For example, in the issues of fraud, banks can go beyond what they are doing currently and work out ways that they can do to reduce fraud penetrated by both employees and outsiders. Employee civic education is important, and this education should encompass both the top management and clerical. This will help to make all staff members aware of what they ought to do and what they not do and their implications. Banks can also now and again try to maintain good software and hardware that is updated to compete in the market. This will in turn improve banks' reputation and people will then start to trust such banks and customers will now be freely able to transact on digital platforms since they are assured that things such as frauds are things of the past.

2.7 Digital banking services from the customer's perspective

This section looked at the benefits that customers enjoy in the times of online banking, but also the section looked at the challenges that customers face in the efforts by banks to accelerate the growth of digital banking in the country.

Digital banking has presented an opportunity to customers in that, they can have access to capital through digital banking (Al-Ajlouni & Al-Hakim, 2019). Customers open accounts with banks for different reasons such as for safe custody of their money and to also have access to loans to boost their lives and their businesses. Digital platforms have provided platforms where customers can access loans right in the comfort of their homes or offices. The banking system has been enhanced in such a way that the system can scrutinise and approve loans according to the customer's acceptable amount allowed. This has provided an opportunity for customers to have access to financial assistance even though they are living in remote areas. Digital banking has provided a platform for most of the customers to be self-reliant and financially stable.

Secondly, digital banking has also provided a platform for financial inclusion in the sense that, digital finance has improved access to financial services to underserved groups (Al-Ajlouni & Al-Hakim, 2019). For example, in Malawi, most of the people live in rural areas unlike those who live in urban. It is usually these people who live in rural areas that are neglected in terms of access to financial services because of their geographical location. But with the coming of digital banking, these people are now being included in the financial inclusion drive since it only takes one to have any kind of a mobile phone to have access to digital banking such as mobile banking. The Reserve Bank of Malawi is also advocating for a cashless society where people need to make conscious decisions to be involved in at least one of the digital platforms that are being offered on the market (Reserve Bank Of Malawi National Payments Report, 2022) This will lead to a digitally conscious society where access to financial services such as loans will be available despite the geographical location of a person.

Digital banking has also provided an opportunity for customers to have tailor-made banking services that are suitable according to their preferences (Al-Ajlouni & Al-Hakim, 2019). As has already been alluded to in the introduction, the younger generation is the most involved in digital services activities because of their way of living; they want services that are flexible and suit their modern lifestyles. With digital banking, it is possible to have tailor-made services that are specifically unique to a person. Banks can also take advantage of these tailor-made banking services to have a competitive advantage over their customers. Usually, customers flock to banks that will provide them with the services that they need. With tailor-made services, banks can use these to their advantage and have many customers flocking for such services which in turn will contribute to the bank's revenue through commissions and charges that are incurred using these tailor-made services.

Despite the many opportunities that digital banking can provide to customers, some challenges are being faced by customers in their execution of transactions of digital banking. Below are the challenges that customers are facing in dealing with online banking.

Firstly, customers are the most hit when it comes to fraud in online banking. External threats such as hacking, sniffing, and spoofing are exposing customers' accounts to risk of being defrauded (Selvam et al., 2018). Bank employees and other third parties have become more advanced in the way digital platforms work. Therefore, they are working tirelessly to find ways how they can defraud such customers. It is because of these fraudulent activities that some customers still resort to traditional banking which is safer for the customers. However, the traditional banks are costly to the banks because maintaining such branches is expensive and the banks also need to have enough manpower to serve such customers. This therefore is derailing the drive for banks to go fully digital.

Secondly, financial illiteracy is also one of the contributing factors to the derailing of improving digital banking in developing countries like Malawi (Ofosu-Ampong, 2021). Financial literacy is lacking in most of the developing countries, as such, for the banks to advance in their digital drives it is difficult since their targeted customers are not aware of the developments that are happening. Because of this financial illiteracy, most of the customers still must spend their time at traditional banks instead of putting that time to good use. It is also because of this financial illiteracy that some of the customers that were migrated into the digital platforms do not know how best they can operate their accounts, as such, it is found that instead of keeping confidential information to themselves, they share that information with third parties who are malicious and who later defraud them.

Studies have also shown that banks do not put in much effort to educate their customers on the good use of these digital platforms; banks will only introduce new products on the market and yet it was found that even employees are not very knowledgeable on how to go about such products. A study by Rahman et. al (2017) in Bangladesh at Dutch-Bangla Bank Limited found that customers do not have enough knowledge about the e-banking services offered by the bank (Ali et al., 2021). Another study by Nam (2014) in Vietnam also found that the major challenge in the adoption of online banking services was the lack of knowledge by both employees and customers (Ali et al., 2021). As such, it is very hard for employees to impart such knowledge to customers since they are also not aware of the online products being served by their banks. Both studies

therefore recommended that management should take the initiative to teach their staff members how they can go about digital platforms for their personal use which they can later teach the customers. Staff members are the first line where customers first meet them. If they (staff members) lack such important information to tell the customers, then it is hard for customers to be kept abreast of the knowledge of digital banking products and services.

A study by Farooqu and Rajani (2017) on Indian banks also found that customers are not willing to take part in online banking because of the security risks, trust factors and privacy risks associated with online banking (Ali et al., 2021). It was found that even customers who have never been r=defrauded were reluctant to take part in online banking because of the speculations that they heard from their fellow counterparts. This assertion that online banking has a security risk goes back to the lack of knowledge that people have concerning online banking. This therefore influences people's attitudes not to use digital platforms.

As we are living in the digital age, online banking will always affect us positively or negatively in one way or the other. Online banking is booming globally, therefore, developing banks also need to take a step further in the way they conduct their businesses to match the global standards. This section has reviewed literature from both developed and developing countries and how digital banking is positively and negatively changing them. Different reasons have emerged that still need to be looked at such as attitudes and beliefs that are affecting the use of digital banking in the developing. This research therefore went further to find out the challenges that both customers and banks face in the adoption of digital banking in the country. The results from this study would help to shape the future of digital banking in the country which also aligns with the country's vision 2063 strategy to have a digitally advanced country by that time.

2.8 Chapter summary

The chapter has provided the Technology Acceptance Theory and the Diffusion of Innovation Theory as theories that influence the study. It also discussed the concepts that inform the theory and the relevance of the theory to the study. The researcher also reviewed the existing literature on the adoption of digital banking. The review has demonstrated that there is a need for more literature on how developing banks can best attain digital banking acceleration in Malawi, and this study aims to fill this gap.

CHAPTER THREE: RESEARCH METHODOLOGY

3.1 Chapter overview

This chapter discusses the research methodology. It presents the research philosophy, research design, population, sampling technique, sample size, data collection, data analysis, ethical considerations and validity and reliability of the results.

3.2 Research Philosophy

This research was guided by Interpretivism research philosophy. This philosophy considers humans as different from physical phenomena as they create further depth in meanings with the assumption that human beings cannot be explored in a similar way to physical phenomena (Alharahsheh & Pius, 2020). The theory is also based on the assumption that social reality is shaped by human experiences and social contexts (Alharahsheh & Pius, 2020). Since this research was qualitative and needed to hear views and experiences of customers from the use of digital banking services, Interpretivism was there used to guide the research.

3.3 Research design

This study employed a qualitative research method. Qualitative research is defined as the type of research method that is concerned with the deepening of understanding of a given problem (Almeida et al., 2017). The objective of qualitative research is to produce in-depth, illustrative information to understand the various facets of a problem under analysis (Crick, 2021). Maxwell (2013) advocates that qualitative works with the universe of meanings, motives, aspirations, beliefs, values, and attitudes which respond to a deeper space of relationships, processes and phenomena that cannot be reduced to the operationalisation of variables (Almeida et al., 2017). Since the study was concerned with understanding the challenges that both banks and customers faced in the times of digital banking and explored the growth strategies that were there in digital banking, cross-sectional research design was best used for this research as it provided a platform where the researcher was able to ask questions from both parties and sought to gain an understanding from the parties on the subject that was under discussion.

3.4 Population

3.4.1. Study setting

The study was conducted in the southern region of Malawi, and it included the service centres that were located in Blantyre. This was due to the researcher's proximity but also because there were quite a substantive number of banks that the researcher took advantage of.

3.4.2 Characteristics of the sample

This study had participants from both the customers and the bankers themselves. Customers who were registered for at least one of the digital banking platforms were randomly sampled to include them in the study. The total sampled number of customers for this research was 20. The study also included four participants who were managers from the Digital Financial Services (DFS) department, one from each bank as these are the people who interface the digital banking platforms, and they are aware of the most challenges that customers face since the service centres (branches) report the challenges to them. These are also the personnel that are involved in the strategic planning and development of the digital banking platforms.

3.4.3 Sampling technique

For this study, different sampling techniques were used to have the desired sample. Purposive sampling was used to choose the banks that would take part in the survey. First, the researcher looked at all banks in the country that have been on e-banking for at least five years and there were eight banks that fell into this category. It was found that eight banks have been on e-banking for that period. From the eight banks, the researcher did a simple random sampling to come up with five banks whose customers and staff members (from the Digital Financial Services department) were to take part in this research and the following were the banks that were sampled: National Bank of Malawi, Standard Bank, FDH Bank, NBS Bank and Ecobank. Five Digital Banking Services staff members, one from each bank, were also purposively sampled because they are the key informants in terms of the digital innovations that are happening in the banks and because they interface with the service centres in terms of the challenges that may arise now and again. The customers were equally divided to represent their respective banks that they bank with; four customers each from the five banks took part in the study. During the interviews, the customers were conveniently sampled from the service centres depending on their availability, willingness, and convenience.

3.4.4 Sample size

Unlike quantitative research design, the qualitative research method usually has a small sample size. This is the case because the researcher gets more rich information from the participants and the researcher must make sure that the information is a representation of the ideas being advanced by the participants. A sample size of one to ten is recommended in qualitative research according to Mocanasu, 2020. Other authors advocate for a sample size of 13 to 15 while others advocate for a sample size of 20 to 25. For this study and to get rich information, this research had a sample size of four participants for the banks' key informants from the five banks that were sampled (one bank did not grant an interview for their key informant). And a sample size of 20 customers was used for the study; four customers each from the five banks that were sampled took part in the research. Even though a sample size of 20 customers was used, the researcher still relied on saturation levels to reach. For the customers, saturation was reached at 20 as there was no new information that was being unveiled from the interviews. For the banks' key informants saturation was not used as the information from the respective banks was unique in its own way. A total of 24 participants; for both customers and banks' key informants was used.

3.5 Data collection

This research used interview guides to collect data from the participants. Interviews give the most direct and straightforward approach to gathering detailed and rich data which is useful for a particular phenomenon (Aspers & Corte, 2019). Interview guides allow flexibility in the structure of the questioning to allow the interviewee to respond promptly to what is relevant and ask for clarifications from the researcher (Adosi, 2020). The interview also gives room to the researcher to make use of probing questions in the guide to make the information gathered more relevant and useful for the topic under research (Adosi, 2020). Since this research explored the challenges that the banking sector is facing in accelerating digital banking and the strategies that could be explored in digital banking, interviews gave more room for the respondents to be able to explain their answers in a manner that was beneficial to the researcher's work. The researcher was also able to ask probing questions that followed up on the interactions that she had with the respondents.

The customers were mainly asked to narrate the challenges that they were facing for them not to fully participate in e-banking services, their knowledge of the available e-banking services and what hindered them not to taking part in those available e-banking services. The DFS staff are the ones who implement online banking at the branch level. The researcher also found out what

challenges they faced in their delivery of online banking at the bank level. From the challenges that they faced; they were asked to narrate the strategies that they foresaw if the challenges that they talked about were going to be resolved. The four DFS staff that were sampled for this research were also allowed to talk about the future of online banking from their perspectives on the way online banking was being implemented in the country.

The interview guides were framed in both closed and semi-structured questions. Semi-structured questions are loosely structured to give participants more opportunities to express themselves freely (Adosi, 2020). This helped to elicit more information from the respondents since they were not confined to closed questions only. This also helped to reveal the attitudes, beliefs, preferences, and ideas that the respondents had towards digital banking in the country. The online banking platforms that formed the basis for this research were; internet banking, mobile banking and debit cards as these are commonly and widely used by personal customers.

3.6 Data analysis

Data analysis transforms data into findings by bringing order, structure and meaning to the mass of collected data (Patton, 2002 cited in (Adosi, 2020). This research was guided by thematic analysis following Braun and Clarke's stages for qualitative data analysis.

Thematic analysis is a method of identifying, analysing, and reporting patterns or themes within qualitative data (Braun & Clarke, 2006). This method undergoes six stages for the researcher to analyse the data and come up with meaningful results. The first stage is whereby the researcher familiarises herself with the data. The researcher needs to be familiar with the depth of the content and search for meanings, patterns, and other important points. The data is also transcribed at this stage for the researcher to familiarise herself with the data. The second stage is where the researcher searches for codes from the data. Codes are elements of raw data which can be assessed in a meaningful way regarding a phenomenon (Boyatzis, 1998 cited in (Braun & Clarke, 2006). Coding can be done by the researcher highlighting important information that she sees as useful for the research. The third step is whereby the researcher searches for themes from the data. This stage involves sorting different codes into meaningful themes. After this stage, the researcher revises the themes which have been formed, and at this stage, this is where the researcher can change the names and come up with more meaningful ones that will communicate with the research questions at hand. The next stage is defining and naming themes. This stage involves formulating exactly what we mean by each theme and figuring out how it helps us to understand the data (Braun &

Clarke, 2006). The last stage of thematic analysis is where the researcher writes up the discussion of how the thematic analysis was done. In the write-up, the researcher narrates how the data was collected and discusses the findings that have been made from the data that was collected.

During the second stage of thematic analysis which is coding, the researcher inputted the data into qualitative software called NVivo 12 plus. This is a qualitative data analysis software that helps to organise and analyse qualitative data. Once the data was fed into the NVivo 12 plus, the subsequent steps of thematic analysis as highlighted above were followed to analyse the data. The researcher analysed on her own.

3.7 Ethical consideration

This study considered all ethical considerations in order not to offend the participants who helped to get the responses. The researcher got consent from the participants before the research was conducted. The participants signed on to the interview to show their consent to the research. The respondents were made aware of the right to withdraw at any point in the research whenever they felt they could no longer continue with the study. To ensure anonymity, participants' names were not disclosed, instead, numbers were used. The participants were also made fully aware of the nature of the study and that participation was strictly voluntary. Lastly, the participants were given adequate time to ask questions and present any concerns if they had any. This helped the participants to feel that they were part of the research and that whatever information they were giving out was important to both the researcher and the respondents (Mohd Arifin, 2018).

3.8 Validity and reliability

Reliability assesses the consistency of results over time while validity is the confidence with which conclusions can be drawn from an analysis (Hafeez-baig & Chakraborty, 2016). To have valid and reliable results that will be a representation of the study findings, the researcher did the following to achieve validity and reliability; firstly, the researcher took the interview guides to the experts in the banking field for them to assess whether all the areas that needed to be looked at were addressed in the guides. The need for amendments was advised by the experts to have rich data that would come out from the research. The researcher employed one data analyst to compare the results to reduce biases. The analyst worked on the data independently only to merge the findings (themes) with the researcher later and checked for similarity rate which, using NVivo, the Kappa coefficient for nodes and sources was 0.78. Further, the researcher herself made sure that all necessary protocols for the research were observed.

3.9 Chapter summary

This chapter discussed the research methodology. It has presented the research design, population, sampling technique, sample size, data collection, research procedure, data analysis, ethical considerations and validity and reliability. The researcher also provided justifications for adopting each factor. This study used a qualitative method. It had a sample size of 20 participants for the customers that were interviewed and the saturation point was reached at 20. There were four key informants instead of five as one bank did not grant the researcher an audience. The total number for participants was 24; 20 customers and four key informants. Thematic data analysis was done using Nvivo 12 Plus. Furthermore, the chapter highlighted the issues of trustworthiness and reliability in the study.

CHAPTER FOUR: RESULTS AND DISCUSSION

4.1 Chapter overview

The chapter discusses the results of the study. The questionnaire return rate for this research was excellent as out of the 25 participants that needed to be interviewed, 24 granted the interviews and only one key informant did not grant the interview. Four main themes emerged from the data and these themes have sub-themes. The following are the themes that emerged: lack of awareness, banks' complacency, the need for change and external factor challenges. A discussion follows from the themes that were derived from the research.

4.1.1 Demographic characteristics of the respondents

Demographic characteristics of participants in this study are shown in Table 1. There was a balanced representation in terms of participants that came from the banks that took part in this research. However, there was a slight difference in terms of the representation of the gender of the participants; 55% was represented by males and 45% by females, marital status where 65% of the participants were married and 35% were single, age range of 18-30 years was represented by 30%, 31-40 years dominated by 60% and 41 years and above was represented by 10%. The employed were represented by 79%, self-employed 13% and 8% for unemployed.

Table 1: Demographic information of the sample for the qualitative research

Participants groups	Sub-groups	Frequency	Percentage
Gender	Male	13	54
	Female	11	46
Marital status	Single	9	38
	Married	15	62
Age range	18-30 years	6	25
	31-40 years	16	67
	41 years & above	2	8
Employment status	Employed	19	79
	Self-employed	3	13
	Un-employed	2	8

A total sample of 24 (13 males and 11 females) participated in the in-depth interviews which were conducted in the selected banks in Blantyre, namely: National Bank of Malawi, Standard Bank, FDH Bank, Ecobank Malawi and NBS Bank. From the 24 participants, 20 participants were customers from the previously mentioned banks and four participants were the key participants that are in digital financial services from all the banks except for Standard Bank.

4.1.2 Theme and sub-themes

The participants' responses were analyzed using qualitative research data analysis software called NVivo 12 plus where codes were generated, and later themes were derived. The analysis came up with the following four overarching themes: i) lack of awareness ii) banks' complacency iii) the need for change and iv) external factors and challenges. Table 2 illustrates all the themes and their sub-themes for each theme with their extracts from the data.

Table 2: Themes and sub-themes

Theme	Sub-themes	Example of an extract	
Lack of Limited knowledge of e- awareness banking services		•	
	Limited use of e- banking services	I usually make bill payments and airtime buying, that's all	
	Lack of cyber-security knowledge	The security measure that I know is just as simple as not giving someone my pin, yeah, I think that's it	
	Limited sources of information	Usually the bank itself communicates by sending messages.	
Bank's complacence	Lack of sensitization	It depends on how we get that information because the bank doesn't inform you directly. Its either you have to go to their Facebook page for you to know more of certain information. If you don't follow their Facebook page it means you will not have that information. So maybe if the bank would have a different way of trying to communicate to their customers.	
	Lack of urgency in serving customers	I should mention, my phone number for some reasons, has never successfully received notifications from the bank. It's the problem that has been there for years, I reported it multiple times but it was not resolved so I ended up giving up.	
	Increased application downtimes	Usually, when the network is down, and you are trying to do a transaction, let's say over the weekend, the system is not able to return that money to your account. You are forced to wait until the next working day, which is a Monday. And then usually when you are following it up on Monday, they tell you 'Give us 48 hours.' I'm usually not pleased with that	
External factors challenges	Customers' rigidity to change	The other challenge is, as much as we have seen an improvement in the adoption, but we still have a challenge especially with some customers who as much as we school them, they are used to the old traditional banking. They haven't yet adopted the digital technologies and the like. So, it still poses as a challenge because for us, our aim is to make sure that we simplify the way they do transactions	
	Third party technical challenges	When we are offering these solutions, we partner with other players. For example, for electricity we partnered with ESCOM. So it might be that at the point the customer is making the payment, ESCOM system was down. Obviously the token can't be generated. So you find that one has paid but the token is not there. So as a bank, we take it up on	

		ourselves to check in our systems and with our partner too. If the token is retrieved, we provide it to the customer. If it is not there, then we refund. But in such instances it is the third party's system issue that you can go beyond	
The need for	User-experience	In other markets I have seen people use the USSD to pay for goods in a shop and things	
change	enhancement	like that. So it's something that we'd love to see because that is where the convenience	
		comes in. We want to see people not carrying cash but using the different digital platforms	
		that are there.	
	Continuous feedback	I think banks should also listen to customer's feedback more and see where they can	
		improve on their services and the processes underlying the e-banking platforms so that they	
		are more user friendly.	
	Need for introduction of	First one maybe would be the KYC one. I would expect them to improve from what happened	
	automated services	this year. A lot of customers complained especially the bank which I bank with, it has a lot	
		of customers and there are a lot of complaints too. I expect them to use the complaints that	
		were there to improve the KYC process	

4.2 Customer's knowledge and user experience of digital banking service

4.2.1 Lack of awareness

A lack of e-banking awareness was the dominant theme throughout the research. It was found that customers were not aware of how some of the e-platforms work because they lacked sensitization from the banks themselves. The banks are quick to introduce new products or services without first sensitizing the masses on how that service works. It was also found that customers got the information about new services or changes from relations or friends in cases where they indicated having heard about something new that has just been introduced. Lack of cyber security was also prevalent, as out of 20 participants that were interviewed, only six had high knowledge of cyber security. This theme, therefore, was supported by four sub-themes, namely, i) limited knowledge of e-banking services, ii) limited use of e-banking services, iii) lack of cyber security knowledge and iv) limited sources of information.

4.2.1.1 Limited knowledge of e-banking services

During the research, it was found that most participants had limited knowledge of e-banking services. Customers attributed e-banking to the act of sending funds from their mobile gadgets to another gadget. Other participants were not aware that ATM cards, and credit cards, are all regarded as e-banking platforms, and yet, they were so inclined to use mobile phones as their main source of transacting on e-banking. Consider the following sentiments from a participant who translated what e-banking is:

Mmmh, what I know about E-banking is just how you can do transactions from the bank like to your phone or you can make payments through your phone, yeah that's all I can say (customer in-depth interview 1).

It is because of this limited knowledge of e-banking that many people are not patronizing other e-banking services such as Internet banking, and credit cards because probably they are not well exposed to them which makes their knowledge limited.

4.2.1.2 Limited use of e-banking services

It has been found out from the research that there is limited usage of the available e-services; the commonly used being funds transfers and utility bill payments. Out of the 20 participants that took part in this research, 18 mentioned that they use e-banking for funds transfers and from the same

20 when they were asked what other services they use, 16 mentioned that they use e-banking for funds transfers. Customers are also able to pay for rentals, check for forex rates, pay for taxes and make tuition fee payments, but all the above four were once mentioned that they are used. Consider the following statement that a participant mentioned about their usage of e-banking:

Ok, I use e-banking services almost daily. Maybe buying some airtime, paying some bills, transferring money and other things but usually it is the utilities like payment of bills, transfer of money and finding mobile airtime on mobile banking. So, those are the main ways in which I access and how often I do it

As the above extract shows, several other participants mentioned that the above are usually the types of services that they use. The limited usage of a few e-banking services can contribute to congestion in the banking halls as more customers flock into the banking for transactions that they can do on their phones. The lack of knowledge is probably what is pushing them into the banking halls.

4.2.1.3 Lack of cyber-security knowledge

Knowledge of security is one of the concerns that has been highlighted by the participants in the in-depth interviews. During the interviews, participants were asked about their knowledge on how to keep their accounts safe, only six participants out of the 20 showed that they had high knowledge on security issues. An issue of customers being duped by fraudsters because they (participants) gave out personal information that is meant for them was also mentioned. Fraudsters have come up with tactics where they pose as if they are from the bank and ask customers to give their personal information such as personal identification number as part of updating their accounts. Customers end up giving out such information which results in fraudulent activities in their accounts. Consider the following sentiments from a key informant from one of the banks that participated:

"You find that in some cases customers do share their pins. They share pins for their ATMs, they share passwords for their internet banking, and they share credentials for mobile banking, be it knowingly or unknowingly" (key informant in-depth interview 1)

As much as it is a shared responsibility between the banks and the customers, the banks also need to take an active role in making sure that their customers are well-sensitized and equipped on how best they should keep their information safe. Consider another sentiment from a participant about how he was duped without the bank's knowledge:

The worst challenge I've ever encountered was this fraud thing. Somebody managed to make online payments without me knowing. They were paying for some things online that I didn't know. I was just getting notifications like 50-something thousand gone, and like 15 thousand something gone and the next day it also happened until I had to get my account blocked (customer in-depth interview 8)

Such malpractices tend to push away customers from using e-banking to traditional banking for fear of getting their funds duped. It is disheartening also to the customers to note that such malpractice would happen to a customer two to three times without the bank's mechanisms of tracing such malpractices.

4.2.1.4 Limited sources of information

It was found that during the research, limited sources of information were one the key issues that was keeping customers away from what was going on. From the twenty participants that took part in this research, most said that they get information the text messages while the second most said that they get information from social media. As much as social media might be a powerful tool for the younger generation, people still consume newspapers and radios for information. The radio is one of the powerful tools for communication as it reaches out to the masses, but it is sad to note that banks no longer use this medium to reach out to the masses. Consider the following sentiments from a participant:

They usually share some alerts, like via email and sometimes via SMS and we also have

Instagram and Facebook platform where they usually constantly post the recent developments

(in-depth interview 15)

From the time of the research, banks were paying attention to two or three sources of communication and neglecting the others which are also powerful in reaching out to the masses. It is therefore recommended that the banks use all types of communication so that they are sure that they are well covered in terms of communication that reaches out to their targeted masses

4.3 Challenges hindering growth of digital financial services in Malawi

4.3.1 Banks' complacence

It was established that banks have become complacent while executing their duties. That urgency to attend to customers is no longer there as some other sentiments they get them from people's

frustrations on social media. The same customer issue is being reported more than twice for that customer to be assisted. There has been an outcry from customers that usually at the end of the month the system is either slow or down which makes it hard for customers to transact at such a crucial time of the month. It has now become a norm that network connectivity challenges have now been familiarized by the customers and no lasting solution seems to be provided. Complacency has affected the banking industry as they take serious business as "business as usual". This theme was supported by the following sub-themes: i) lack of urgency ii) lack of sensitization iii) increased application downtimes and iv) lack of accessibility and convenience.

4.3.1.1 Lack of urgency

The lack of urgency was reported by the participants that banks were lacking that urgency to assist them on their pressing issues. The bank officials were reported to not take their issues seriously as on some occasions, an issue took more than a year for a customer to be assisted. With this lack of urgency, some other customers resorted to the old way of banking which was banking or using another eplatform which they were not comfortable with but that was the only option that they had. Consider the following complaint from a participant:

Again, I should mention I had access to Bank X's mobile app but at one point in time, I had challenges logging in. Those challenges have never been resolved; I complained about it to the bank on multiple occasions but a few empty promises, and still now the issue hasn't been resolved. So, I am forced to go to the internet portal but my preference would be the mobile app (in-depth interview 12)

From the above sentiments from a concerned customer, it shows that customers were not getting the services or products that they wanted, they just used the available services for them to complete certain transactions. But if they were given a wide range of choices or if their challenges were being taken into consideration, the e-platform experience would have been a better one.

4.3.1.2 Lack of sensitization

The banks in the country have not taken an active role in making sure that their customers are well-informed about the products and services that they offer on the market. Civic education has not been done in the way it should. This has resulted in customers not taking an active role in digital banking acceleration. Despite this shortfall, the banks' key informants were quick to acknowledge this shortfall

and said they were doing everything possible to sensitize their customers on the need to know and be well acquainted with digital services for the economic benefit of this country. Consider the following sentiments from one of the key informants:

One of the things that we have been suffering the most is fraud. On our end, we try our best to safeguard the customers' money, but the fraudsters are also very clever. They always find ways of requesting Pin or maybe the OTPs from the customers. And sometimes the customers are not aware of these kinds of situations. They send up giving information that they were not supposed to give and then they are eventually defrauded. What we have done is, we have teamed up with the marketing team to roll out campaigns, where we send out SMS alerts alerting the customers that they shouldn't disclose their personal information to avoid being defrauded (key informant in-depth interview)

Unfortunately, fraud has painted a bad picture of online banking, but if the banks are vigilant enough and take an active role to avert these challenges, customers are willing to embrace change as has already been seen by the number of transactions that customers make daily.

4.3.1.3 Increased application downtimes

System downtime was reported by almost all the participants. The participants complained that most of the time they have challenges with the system usually at the end of the month. This is the time that most people are busy with the e-platforms to initiate different transactions, ironically, usually the system is either slow or down such that they fail to complete their transactions. Customers are therefore not left with any option but to go and queue in the banking hall which is a big inconvenience to the customer. This is what a participant had to say about the system downtime challenges:

I would say there are specific times, especially during month end when the application is mostly down. If you are trying to send money or let's say you are in a shop you are trying to make a payment on the POS machine and boom, there is no service! I rarely get joy when I am trying to transact, and the application is down (participant in-depth interview 6)

These connectivity challenges have been there for some time, a certain participant complained. They lamented that the banks are aware of these challenges and probably they have become complacent in taking an active role in making sure such challenges are resolved once and for all.

4.3.1.4 Lack of accessibility and convenience

Online banking is supposed to be convenient and accessible to all who want to indulge in it, but it has been shown that online banking accessibility is somehow limited. It has been revealed from the indepth interviews that customers were not able to do all the transactions that they needed because some services were not available on the platform. It was also revealed that the banks and the MNOs sometimes had disagreements that made it impossible for other services such as money wallets not to be on the list of some mobile or internet applications. E-banking has also not solely provided the convenience that people need to comfortably transact at the comfort of their preferred locations. Customers are still left for example to go to the banking hall to apply for forex when this can be automated. Consider the following sentiments from a participant:

The limitation in terms of the services you can get on the online platform is discouraging. There are some services which will require you to have a face-to-face meeting with your banker, so we are not yet there, but maybe we will get there. For example, if you want to apply for forex, you must physically go to the bank and meet your banker which is time-consuming (participant 20 in-depth interview)

No automation of services has created an environment where customers are being forced to queue in the banking to make transactions and if the banks were to take a leading role, the issues that are being faced now would be history.

4.3.2 External Factor Challenges

Banks do not work in isolation, other players in the industry complement the works of the banks. Some of the challenges that the banks are facing could be attributed to the negligence or system challenges that their third parties are facing which cripple service delivery for the banks. As much as the banks are somehow complacent in their service delivery, with some other issues they cannot do anything, let alone leave those challenges to the owners until such a time when the third party resolves that issue. This theme, which falls under objective two, was supported by the following sub-themes; i) customers' rigidity to change and ii) third parties' technical challenges.

4.3.2.1 Customers' rigidity to change

Customers are key players in the drive for digitalization. All the efforts that banks are involved in revolve around a warm good user experience for their customers. But it is the nature of man, there are different stages that one takes to accept an order. Customers, as the backbone of any business, are also somewhat derailing digital growth because they are unable to embrace new change. Consider the

following statements from key participants on the customer's conduct regarding the use of POS machine:

So, you find that for example, a shop, you have some shops that are still focusing on cash transactions while as banks, we are offering a digital solution for their businesses. So, you provide a solution, for example, you would give a POS machine to a customer for them to accept card payments. But now you find that those merchants hide the machines, the reason being maybe they are running away from the commissions that we bank charge on them.

Such external challenges have an impact on the adoption and growth of digital banking as banks also rely on the players to actively participate when new initiatives have been introduced to accelerate digital growth.

4.3.2.2 Third parties' technical challenges

Challenges with infrastructure surfaced as the banks complained that they were unable to deliver what they had because the third parties did not provide a good conducive environment for them to thrive. This is what one key informant had to say:

As much as it is a bank's system issue, it also goes back to the partner. Because when you are offering these solutions, we partner with, if it is electricity, we partnered with ESCOM. So it might be that at the point the customer is making the payment, the ESCOM system was down. The token can't be generated. So, you find that one has paid but the token is not there. So as a bank, we take it up on ourselves to check in our systems, to check with the partner if the token is there then we provide feedback to the customer.

With such a challenge, it is difficult to give a customer some concrete as the banks are also waiting to get feedback from their partners. It is therefore that both banks and partners have a common understanding of how best such issues should be handled to give an excellent service to their customers.

The issue of vandalism was also mentioned by one key participant that this issue is commonly affecting their services. Since the banks rely on mobile phone operator's networks for the systems to work, vandalism is affecting the work of the banks as well. Consider the following sentiments:

This one is two-fold, it's two ways; it can be a bank issue, or we have had instances where some people have torn fibre cables somewhere and you find that we banks don't have any

connectivity and we can't render digital services to our customers (key informant 3 indepth interviews)

Another key informant also complained that the country's infrastructure leaves a lot to be desired; the government is not doing enough to make sure that different players, banks inclusive, are given a good playing ground where transactions are done without challenges. This is what one key informant had to say:

Our infrastructure is not yet to the level where the services are available 100 percent, so there are incidences that the customers want to transact but then you hear that there is a fibre link cut somewhere and there is no network or there is no internet. We have cases whereby we, have power outages and the systems cannot operate (key informant 1 in-depth interview)

Challenges such as vandalism or power outages are affecting the services of digital banking. In such cases, banks cannot do anything as such responsibilities are in the custody of others.

4.4 Banker's and customer's perceptions of the growth of digital banking services

4.4.1 The need to embrace change

The need to embrace change was one of the themes that was highlighted in this research. Although customers expressed challenges that they are facing in the adoption of digital financial services, there are statements that they have made to show that they want this change now than before. It is also interesting to note that despite the challenges highlighted in the interviews, customers expressed satisfaction with the current performance of digital banking. Out of the 20 participants, 70% showed high satisfaction with the performance of e-banking platforms. This theme was supported by the following sub-themes; i) user-experience enhancement and ii) continuous feedback.

4.4.1.1 User-experience enhancement

Customers are longing for a user enhancement that is friendly and able to meet their demands. It was noted that some customers complained that e-platforms are not user-friendly, as such, they are not able to use them fully. System challenges have also contributed to customers not having a good user experience. It was noted that sometimes a customer would initiate a transaction, but that transaction does not materialize due to network challenges that interrupt such transaction. Participants also complained that they had challenges in receiving alerts to confirm that their

transaction was successful and others also complained that they were unable to retrieve utility tokens due to intermittent system failures. Consider the following sentiments from concerned participants:

The first thing that comes to my mind is in terms of payment of utility bills. Sometimes you will pay for a utility bill, and you don't get to receive the token. Maybe let's say you are buying electricity then in the end you are not getting the token for the electricity, that's where the puzzle comes in "Should I contact the bank?" "Should I contact ESCOM?" You are not sure who to contact (participant in-depth interview 4).

In such scenarios as the above customers are left puzzled as to who to contact as they do not have proper information of who to get hold of in such scenarios. This leaves customers with no choice but to abandon e-platforms and revert to the old way of transacting as they no longer trust the platforms.

In the same vein of crying out for solutions/help, participants have also gone ahead to provide solutions to the challenges that they continue to face to have a good banking experience. We are living in a time when collaboration with other service providers is so crucial. With an influx in the number of people that have access to cell phones, whether smartphones or not, this has created room for expansion and an opportunity to reach out to the unbanked. Banks have now started to collaborate with Mobile Network Operators (MNOs) to have a link where a customer who has a bank account can transfer funds to a mobile wallet and vice versa. As much as this is the case in some banks, some banks are not in collaboration. And for those banks that have already made these agreements, customers are having challenges in terms of the delivery of these services. Consider the following sentiments from a participant:

I look at the e-banking future in a way that it will even be smoother than now, transactions will not take time to go through and also, and they will be well-coordinated with mobile money services. Sometimes people want to access money but maybe there is no ATM nearby so transferring money from the bank to the mobile money wallet will make it easier and more convenient (participant in-depth interview 2)

If the banks and the MNOs can organize themselves well and improve on the grey areas that are currently available, then as a country, the issues to do with failed transactions would minimize which will lead to improved customer satisfaction. The collaboration will also help to reach out to

other customers who were hesitant to be a part of this transaction which in turn can facilitate deposit mobilization for concerned agents.

4.4.1.2 Continuous feedback

It has been noted that participants lack feedback from their banks on the developments that are happening. Participants complained that a certain product or service would just be introduced, and they get to hear about it after some time. It was also noted that customers are not able to get feedback on lodged complaints or enquiries which has shown a failure on the part of the banks. Lack of urgency in giving feedback to customers has also been noticed in the interviews. This has led to frustrations for the customers as they are no longer willing to lodge complaints or ideas as they know that they are going to get the same responses or no responses at all. Consider the following sentiments from a customer:

Again, I should mention I had access to my bank's mobile app and at one point in time, I had challenges logging in and those challenges have never been resolved. I complained about it to the bank on multiple occasions but a few empty promises later it is still unresolved. So, I am forced to use the internet portal, but my preference is still the mobile app (participant in-depth interview 12)

Of particular issue that I had been involved in was the product of cardless withdrawal not working, and the ATM was not working too. I had to go into the banking hall and there I was also told that I needed to come with a chequebook because of the nature of my account (which is a current account). I needed the money urgently but the urgency in assisting me wasn't there. (participant in-depth interview 12)

With these two scenarios of customers not being helped timely may lead to reputation damage for the banks as word of mouth goes very far. Customers are also most likely to resort to the traditional way of banking as they are assured, they are going to get the service which they are failing to get on an e-platform, which defeats all the interventions to get as many people as possible to online banking.

The issue of high tariffs is also one of the interventions that customers have been crying a lot for the banks to consider. We live in a country where most of the citizens live below the poverty line. For such people, being on an e-platform is sort of a luxury as they need to sort out their basic needs first. And if these people are to be dragged to online banking and they face such tariffs, they would

likely be willing to fully participate online. Despite the conditions that people are living in, they are still crying out to the banks to consider them to reduce the tariffs so that they can be a part of the global financial market. Consider the following sentiments from one of the participants:

We hope to see good coordination between banks and mobile money operators whereby people will not have so many charges on the transfers and people are going to access their money easily. This will make it easier to make transfers of a certain amount without any problems. You will not be bothered to go to the bank as often as sometimes we ought to, we can do everything online (participant in-depth interview 2).

If the banks are to consider revising the prices and collaborating well with the MNOs, we can see an increase in the number of online transactions. Many people are still sceptical about embracing online banking due to the perceived high tariffs.

4.5 Discussion of results

E-banking has come to stay. All the banks globally are advocating for the promotion of digital payments which will enhance economic activities and avoid issues to do with fraud as it is now rampant with the introduction of digital banking platforms. As much as this is the case, some other banks especially those in the third world are finding it difficult to adapt to these changes due to the geographical allocation of the countries. This section will discuss the results that have emanated from the research where the researcher was exploring the customers' e-banking knowledge, e-banking challenges in a Malawian setting and the strategies that banks have lined up ahead to accelerate digital banking growth.

4.5.1 Customer knowledge of e-banking services

Issues of knowledge are paramount if banks are to promote e-banking services. Customers must be aware of the services that are being offered and the steps on how they can initiate a transaction. The research has revealed that lack of knowledge is contributing to the customers' low usage of e-banking financial services. This can be attributed to the rigidity of change where people are more inclined to use traditional banking and not embrace digital banking. As much as customers are now seeking a faster and more convenient technology which offers a more rewarding banking experience (Taiwo & Agwu, 2017), other customers are rigid to change especially the older generation. The older generation are the ones that are more rigid to change because they are so engrossed in the traditional way of banking while the younger generation of customers has quickly adapted to the new technology and is

demanding higher quality and more digital-based services (IMF, 2021). It is because of not embracing the new change that is contributing to the people's lack of knowledge on digital products.

The lack of knowledge of e-products has also been vindicated in this research where customers use fewer e-services than the ones that are being offered on the market. There are a variety of products that are offered by banks in the country such as bill payments, funds transfers, tuition fee payments, MRA payments, loan applications, DSTV payments, MASM payments, stop payments and other services. But it is sad to note that out of all the services that are offered, only the funds' transfer option and balance enquiry are the dominant services that customers use. This means that for other services that people may also use such as making a payment to MRA, a lot of people still queue in the banking hall to make such transactions. This is defeating the whole idea of making seamless e-banking where people need to patronize these services. On the other hand, the blame should also go to the banks themselves as it has been vindicated that banks do not relay important information to their customers such as new changes to the e-platforms and the introduction of new products or services. Customers sometimes hear important information from their friends and family which may be misleading. Banks have an important role to play in making sure that they provide useful information to their customers for them to stay abreast of the new information.

A study by Rahman et al (2017) in Bangladesh on Dutch-Bangla Bank Limited showed that customers did not have enough knowledge about the e-banking services (Ali et al., 2021). In a sample of 100, most of the participants showed that they lacked knowledge information on e-banking which was a contributing factor to the low e-banking usage in Bangladesh. Similar research conducted by Beatrice (2017) in Kenya also revealed the same challenge of lack of knowledge in customers which is making it hard to penetrate the e-banking usage (Ali et al., 2021). Customers are not knowledgeable enough to make decisions on the usage of internet banking, as such, internet banking usage is still low. It was therefore recommended that banks should make it a point to train their staff members on the knowledge of e-banking so that they should be able to pass on adequate and useful information to the customers. Banks also need to take a leading role by sensitizing their customers on the usage of Internet banking.

The research also revealed that the banks are not taking a leading role in sensitizing their customers on internet banking usage. Customers complained that when the banks introduce something on the market, it takes longer for them to get to know such information. As much as we are living in the digital era where most of the communications are done using social media and emails, banks also need

to be aware that not many people consume these platforms. Therefore, banks have a role to play in financial inclusion to get the unbanked masses into the banking system. Financial inclusion is defined as "the process of ensuring access to financial services and timely and adequate credit needed by vulnerable groups such as weaker sections and low-income groups at an affordable cost" (Nkuna et al., 2018).

The government of Malawi through the Central bank is also advocating for the campaign of financial inclusion where the rural masses have to be targeted for them to be also part of the financial activities that are happening in the country (Nkuna et al., 2018). The banked population currently stands at 19% (Reserve Bank of Malawi, 2020) which leaves 81% of the adult citizens unbanked. The Central Bank is also advocating for a cashless society where business persons are being encouraged to have different forms of payments where customers can choose from (Reserve Bank Of Malawi National Payments Report, 2022). If these government efforts together with the commercial banks' efforts of financial inclusion, the literacy levels can go down which will see an improvement in people's knowledge of e-banking services thereby increasing the number of people accessing the e-services.

4.5.2 Challenges to E-banking growth strategies

There is an outcry from the customers on an improved user experience of e-banking services. The participants who took part in this research were very open and suggested solutions that would help alleviate the challenges that the banking industry is facing in terms of digital banking transformation. As much as most of the challenges that were highlighted were pointing the blame on the banks, the bankers also had their say on how they envision the future of e-banking in the country and what strategies they foresee that would improve the e-service delivery. The section will discuss the growth strategies that both participants and key informants put across.

In this research, both the participants and the bankers recommended improved infrastructure to support the current e-platforms. The current infrastructure does not support the current demands and a growing number of people who are longing for an improved digital banking experience. Both parties called for the government to act quickly on this situation. Mwiya et al. (2017) recommended that if developing countries are to excel globally, there is a need for a good infrastructure that would propel more people to engage in e-banking service use. Cameroon is one of the countries in developing countries that is doing well in its implementation of e-banking adoption. The country effectively led the reform of its banking sector after the financial crisis and devaluation of the XAF currency in 1994 (Tangakou & Mba, 2015; Kuikeu, 2021 cited in Kala Kamdjoug et al., 2022). Cameroon alone has a

network of 47% of the banking network of the Economic and Monetary Community of Central Africa (CEMAC) (Kala Kamdjoug et al., 2022). Cameroon is one of the leading countries in e-banking adoption in developing countries because of the government's commitment to improving infrastructure. If the Malawian government is to follow the same that other leading developing countries are doing in terms of e-banking adoption strategies, this would improve the service delivery of e-banking.

Security is one of the challenges that has affected the e-banking adoption in the country. Security lapses were noted by both the participants and the bankers. It was reported that customers were being duped of their funds through online transactions that they did not initiate and this affected several customers. Security lapse is a challenge that most developing countries are facing. According to the TAM model, the perceived usefulness of e-banking would only be encouraged if the perceived security is there (Kamdjoug et al., 2022). Once customers perceive that there is no security in the e-platform that they want to use, customers will likely indulge in an e-banking activity. It is therefore advised that banks take an active role in making sure that their customers are well protected from the fraudster's malicious activities. It was suggested by the participants that the use of One Time Password (OTP) should be enhanced to make sure that before any e-banking transaction goes through, customers are notified through a text message for them to give a go-ahead if they know that transaction. Security lapses in e-banking also pose a challenge to the reputation of banks' once customers know that Bank X's customers are not being protected from fraudster activities, it is very unlikely that more customers would flock to that bank. For the banks to protect their reputation, they need to be vigilant enough to stay ahead of the fraudster's activities.

The lack of sensitization was also noted to be one of the challenges that banks were facing in the adoption of digital banking services. Most participants complained that banks do not notify them when new products have just been introduced or if there are any new changes in terms of digital banking. Participants lamented that sometimes they heard of new developments from their families or friends, others also came to know of the new developments on their own when they were navigating through on their e-platform. Customer awareness and product/service knowledge are other factors that were noted to influence digital banking in Nigeria (Festus & Amaechi, 2019). According to Rodgers and Shoemaker (1971), "consumers go through a series of processes in knowledge, conviction, decision and confirmation before they are ready to adopt a new product or service" (Festus & Amaechi, 2019). Therefore, for the successful adoption of Internet banking in the country, there is a need for the banks to sensitize their customers on the new product offering, and how it works, iron out all security

concerns and let the customers know how that product adds value to their digital banking journey. If the banks were to take sensitization seriously before embarking on the release of a product or service, probably the issues that are being faced in the digital banking industry would be lessened.

Banks do not work in isolation as lamented earlier; they rely on the services of other third parties to deliver seamless services to their customers. For example, banks rely on the MNO's availability of the Internet to get their services delivered to the customers. The same banks also rely on the same MNO for mobile wallets transfers between customer's bank accounts and their phone numbers. The banks also have different Memorandum of Understanding between utility companies for the banks to collect utility bills on behalf of the utility companies. However, it has been noted by some of the participants that sometimes when they buy tokens, it takes time for the token to be delivered to their mobile phone numbers. Another participant also complained that it took them more than a year for their DSTV payment to be sorted out. The participant's bank and the concerned utility company were all throwing the blame on each other which left the customer in the middle and frustrated and inconvenienced too. Mobile wallets have a bigger role to play in terms of financial inclusion in the country (Nkuna et al., 2018). If the banks are to coordinate well with the MNOs, the unbanked market can be tapped which can be a benefit to both the banks and the MNOs. The Central Bank is also advocating for a cashless society where the use of cash should be a story of the past (Reserve Bank Of Malawi National Payments Report, 2022). If this initiative is to be achieved, there is a need for both banks and their third parties to have a cordial working relationship which will benefit both the customer and the financial institutions that are involved.

Social media has taken centre in recent times. People nowadays spend much time on social media, and they get to hear more information using social media. In the United States of America, as of February 2019, at least 72% of the adults were on social media, and the number increased to 90% for US adults between the ages of 18 to 29 (Hunt et al., 2021). The younger generation is the one that consumes more of social media, unlike the older generation. To stay competitive, more banks often offer value-added services to make the services more appealing and attractive to their targeted consumers (Taiwo & Agwu, 2017). Since customers now spend most time on time, the participants also asked for e-banking where the platform should also be available on their social media platforms. Instead of logging out of social media to transact on an e-platform, participants asked for seamless digital banking where both social media use and digital banking use can all be happening at the same time. This will create relevance and convenience for the customers thereby increasing digital banking use. While this is the case, participants also asked for more services to be available on digital platforms

where they do not need to travel to the banks for transactions that they can be able to do on their phones. The banks need to enhance the security of these platforms to create a secure digital platform use.

As the adoption of e-banking is increasing, there is also a need for banks to understand the internal business within the organization to make informed decisions that will help in the operation of the banks. One key participant highly recommended the use of data analytics in the digital era. Data analytics is defined as the collection and analysis of large volumes of structured and unstructured data to create value for the company (Cyril & Cyril, 2019). Data analytics can give more insights, predict risk and revenue and make informed decisions for the company (Cyril & Cyril, 2019). With the advancement of e-banking, more transactions are made, as such, it is not easy for the banks or companies to track well the areas that the company needs to concentrate on. With the employment of data analytics, banks would be able to know which platform brings them more revenue, help them in credit risk scoring for their customers and also help in segmenting well their customers to give them a custom-made service which suits their needs (Taher, 2018). It is therefore imperative that the banks in the country take this route for them to deliver services that are meaningful to the customers and this will create value for both the customers and the banks.

4.5.3 Perceptions on growth of digital banking services

4.5.3.1 Customers' perspective

For e-banking to accelerate in Malawi and developing countries in general, there is a need for banks to take an active role in taking into consideration the challenges that their customers are facing in the adoption of e-banking services. Internet banking penetration is low because customers are facing different challenges that probably banks are failing to address or probably the banks are not even aware that customers are meeting such problems. This section will discuss the challenges that customers are facing in the wake of the campaign of digital acceleration in developing countries.

Technological challenges have proved to be one of the challenges that are hindering the growth of digital banking financial services in Malawi. During the research, almost all the participants reported that connectivity issues were hindering their usage of e-platform. Customers reported that most of the transactions happen at the end of the month when most of the transactions happen. This was attributed to system overload since the systems take a lot of transactions than normal. Participants reported issues such as failing to get a token when they bought electricity or water bills or when they were buying airtime. Due to these challenges, customers were left frustrated. Ali et al. (2021) cited that connectivity

and infrastructure issues are some of the challenges that are hindering the growth of digital banking in developing countries. A study by Fetu (2019) on Ethiopian banks on the challenges and opportunities of e-banking also showed the same challenges of poor the in developing countries in the adoption of digital financial services (Fetu, 2019). Infrastructure challenges are making it hard for consumers to fully utilize e-services due to intermittent challenges. The convenience of using e-services is no longer there.

Security is one of the areas that has also been tackled much by the participants in this research. Security is paramount if e-banking is to be highly patronized by many. Customers have reported security lapses where they are being duped of their funds by fraudsters. Of is a case of one of the participants whose account was debited with an online transaction which he did not initiate. It should also be noted that the customer had never made any online payment and yet the fraudsters were able to defraud his account. It is also reported that in the United Kingdom in 2015, roughly 130 million British pounds were stolen from online bank accounts through fraud (Revathi, 2019). This posed a great concern to the banks that were involved and it posed a security threat to the customers that their money was not safe. Such incidences also pose a reputation risk to the banks as they are considered not to be trustworthy in the eyes of their companies. Issues of security need to be looked at critically to avert damages that fraudsters have caused. It should also be noted that it is both the banks' and customers' responsibility to make sure that customers are well educated on how to keep their personal information not to be divulged by anyone else and the customers also practice caution on the best ways of keeping their personal information safe, respectively (Devulapalli & Oruganti, 2021).

Lack of urgency in assisting customers was also reported in this research. Customers reported that, whenever they had a complaint and they presented it to the bank, usually the banks took too long to assist the customers. They also complained that they were unable to use certain applications that were self-registered because the bank stopped the self-registration process due to fraudsters' penetration of the system. Another participant complained that his digital application was not working at the time of the interviews, and he had reported the issue to the bank several times, but a year lapsed without him being assisted and he later abandoned the application. A 2013 survey that was conducted by MIT Sloan Management Review and Cappemini Consulting to better understand how businesses succeed or fail in using digital technology to improve business performance, found that most businesses are failing to cope with the demands of the customers and customers' issues are not treated with the urgency that they deserve (Fitzgerald et al., 2013). Business is being done as usual and customers later

feel frustrated with using such technology because of the way they are being treated by the bank employees.

In this digital era, for one to access an e-platform, there is a need to have access to the internet. The use of Unstructured Supplementary Service Data (USSD) is mostly used by people who do not have smartphones but still want to be part of digital innovations. It should be noted that the high cost of internet services is making it hard for customers to stay abreast with digital initiatives. Developing countries, Malawi inclusive, have been affected highly by the high cost of internet charges. The current price of a standard monthly internet package is \$200, less than \$25 the average salary (Mokoena, 2020). This means that for most of the people in the country, it is difficult to have access to the internet as their priorities are to fetch the necessities; access to the internet is regarded as a luxury. Accelerating digital banking in this environment is hard since people must prioritize basic needs.

High tariffs have also been recorded as one of the reasons why participants failed to participate fully in the digital banking platform. Participants complained that banks charged them high tariffs which made them either resort to transacting in the banking halls where the tariffs were much lower or withdraw the whole salary to avoid being charged now and again when they wanted to transact. This poses a challenge to the economic well-being of the people because they are tempted to exhaust the money which they had prepared to take them up to the end of the money. The Technology Acceptance Model (TAM) asserts two notions of perceived ease of use and perceived ease of usefulness (Ma & Liu, 2011). People tend to use technology if they perceive technology to be useful to them or if it is easy to use. E-banking is being faced with the challenge of lack of penetration because of the high tariffs; people do not see the need or the usefulness of digital banking because they get their satisfaction elsewhere, which is traditional banking. If the tariffs were to be decreased, the perceived usefulness would tend to be improved.

A study by Fonjai (2017) in Cameroon has shown that perceived reliability, trust, security and accessibility significantly impact the perceived usefulness of the e-banking adoption (Kala Kamdjoug et al., 2022). People do not see the usefulness of using e-banking when the platforms do not provide them with security, reliability and convenience due to system downtimes. If the banks are to work out these challenges, more people would perceive e-banking to be useful thereby accepting the technology. The banks should also make it more possible to customize the platforms to be user-

friendly so that more people should not have trouble navigating through the platforms. This would thereby lead to perceived ease of use for e-banking.

4.5.3.2 Bankers' Perspective

Customers are on the receiving end of digital services; they are not aware of what happens for them to meet such challenges that they face. Banks, since they are the initiators of these developments, they have their explanation of what goes behind a service to be delivered to the end-users. In the process of delivering their services to the customers, they also face different challenges that are a hiccup to their smooth delivery of the service. The following paragraphs will discuss the challenges that banks face in the service delivery of digital banking.

Just like customers have complained about the issue of connectivity being paramount in their journey of e-banking, banks also reported that they faced the same challenges from their service providers, such as mobile network operators and internet providers. This is an issue that has been reported by all the banks that participated in this interview. The situation usually got worse during the month's end due to the traffic of transactions that had to be processed at that time. Over 50% of transactions are now being done electronically which poses a strain on the available infrastructure (Taiwo & Agwu, 2017). Connectivity is an issue that has been reported by several developing countries in their quest for an improved e-banking service delivery (Kala Kamdjoug et al., 2022). Connectivity issues are one of the challenges that is chasing away customers from e-banking usage. Banks have a big role to play to make sure that such challenges are being dealt with by their fellow business counterparts such as the mobile network operators. Long-lasting solutions must be found instead of a blame game between the banks themselves and the network providers. Banks need to be vigilant enough to make sure that a permanent solution is provided once and for all.

Secondly, banks also complained about poor infrastructure that is hindering their service delivery. The infrastructure that is currently available is outdated and needs to be updated. It is sad to note that the same infrastructure that used to serve fewer customers then is being strained to serve to large population of people who have migrated to the use of digital banking services (Mokoena, 2020). The banks also attributed this challenge to the government's lack of support in making sure that the country has good infrastructure that should facilitate e-commerce transactions. It was also reported that vandalism of network cables has also facilitated the challenges to digital financial services delivery (Rana & Pandey, 2015). Since banks rely on other service players in the industry for them to deliver quality service, it was sometimes found that people had vandalized cables that were meant to

necessitate seamless delivery of digital services. As much as customers would attribute this challenge to the banks, some of the reasons that the banks are facing are beyond them. This goes back to the policymakers to find alternative ways on how best they can safeguard such infrastructure to accelerate the growth of digital banking.

Resistance to change is also one of the challenges that is hindering the growth of e-banking in the country. As much as the banks would do anything possible to make sure that they reach out to their customers on the availability of digital banking platforms, there are always laggards that will not move with the times. The key informants complained that despite reaching out and sensitizing their customers on digital platforms, still, more customers were still not answering this call. According to the Diffusion of Innovation theory developed by Rodgers in 1962, it asserts that, change is always difficult to adapt to, there will always be the early adopters, those that will adopt the new technology once it has been developed, and there will also be others that will take time to adopt such initiatives despite that technology being good (Al-rahmi et al., 2019). Fonjai (2017) in his research regarding e-banking in Cameroon, argued that banks need to increase e-banking security, accessibility and trustworthiness and reduce the cost of e-banking services to encourage customers' attitudes towards adopting e-banking services Kamdjoug et al., 2022). Fonjai (2017) cited these issues because they are some of the challenges that are making customers rigid to change. If these issues can be taken seriously by the banks, then banks will see a change in the adoption of e-banking.

4.4 Chapter summary

The chapter discussed the results that were derived from the interviews with the participants. Through the themes and sub-themes discussed in the chapter, the results highlight the challenges that the banking sector is facing in the adoption of digital financial services. The participants explained the challenges that they are facing, and the bankers on the other hand also highlighted the challenges that they are facing in their quest for a seamless service delivery. As much as the banks are also facing challenges, it has been noted that most of the challenges that the customers are facing can be averted if the banks are to take a bold step in the execution of their duties. Despite that customers are facing challenges; they were so eager and open to state how they envision the future of e-banking in the country. If the banks are to take into consideration the proposed ideas from the customers, e-banking will not be the same in the years to come. Both the customers and the banks have a role in making sure that e-banking adoption is being taken seriously. Later, a discussion of the results was done which

shows that there are still gaps in the adoption of digital financial services in Malawi and needs an urgent intervention of the banks and the policymakers.

CHAPTER FIVE: SUMMARY OF FINDINGS, CONCLUSIONS AND RECOMMENDATIONS

5.1 Conclusion

The main objective of this study was to find out the challenges that hinder the growth of digital banking in the country and seek to find out the strategies that would help to accelerate and adoption of digital financial services in the country. Both the customers and the bank's key informants took part in this research to have a better understanding of both the customers themselves, as the end-users of e-services, and the bankers themselves as the initiators of these developments. It has been found out from the research that most of the challenges that the customers keep facing all point at the banks.

Most of the challenges that customers keep facing have to do with the banks' response to the customers' needs. Customers reported that banks take too long to resolve their issues and customers later lose hope of being helped. Customers also complained that banks do not let them know of new developments which makes their usage low. On the other hand, banks cried out on poor infrastructure that is hindering their service delivery of e-banking. The bankers also complained that some challenges are beyond them because they rely on the services of other third parties for them to deliver a seamless good banking experience. Some challenges are beyond the bank's capacity which is needed for the policy makers to act on. Despite that this is the case, those challenges that are within the banks' scope, need to be looked at with urgency to instill the lost confidence in the customers.

Both parties came up with strategies that would be beneficial for the acceleration of digital banking growth in the country. Some of the strategies that were proposed were improved security on digital banking platforms to safeguard hard-earned money for the customers and sensitisation of the banks' products and services before they are rolled out. This would help to get customers acquainted with the new product on offer which would also enhance customers' ease of use and ease of usefulness of the e-banking products. The use of data analytics was also suggested whereby it would be used to get more insights on the banks' performance and the banks would be able to segment their customers well and give them tailor-made solutions specifically suited for each customer. The data analytics would also help banks to know which areas of focus they should concentrate on to maximise revenue for the banks. If the banks are to go by with has been suggested, the banking industry would be a different one where customers would feel to be part of the global community.

The Technology Acceptance and Innovation Diffusion Theories have been the basis for this research. As the TAM theory focuses on the perceived ease of use and perceived usefulness, the results from the study have shown that this theory was applicable. Lack of awareness of the digital financial services has been seen as one of the factors that are contributing to the non-adoption of digital financial services. As people are not aware of these platforms, the perceived usefulness of these platforms tends not to be there, as such, the digital platforms are not fully being utilised by the majority. The rate of diffusion of digital financial services also tends to be slow as the banks are not making enough efforts in order to sensitise the majority of the digital financial platforms. If the banks were doing enough of sensitization, then the fears that people have of cyber security threats would have been minimized. People need to be civic educated on how they should navigate the digital banking platforms in order to be vigilant enough. Therefore, the theories listed above are directly linked to how digital banking services are adopted by the majority of the customers.

5.2 Recommendations

The following are the recommendations that the researcher came up with from the results of the study: firstly, the banks need to act with ruthless speed and urgency when it comes to customer complaints. Customers are the reason why banks exist, therefore, if the banks do not resolve customers' complaints, they (customers) will end up frustrated and the customers may end not going to that bank again. Secondly, cyber security is of paramount when it comes to digital banking. Most of the participants complained that the banks' cyber security is weak, therefore, they end up shunning some services for fear of being duped. The banks need therefore to invest in cyber security in order to regain the trust of their customers.

The banks also need to embark on sensitization campaigns in order to reach out to the majority of their customers. It has been noted that some campaigns that the banks embark on do not reach out to their mass customers, therefore, there is need to use media platforms and other channels that will reach to the masses. The banks also need to take on board proactively the financial inclusion campaigns that are being advocated by the Central Bank. There are a lot of people that are unbanked, hence, banks need to take these initiatives seriously in order to reach out to the many Malawians that are unbanked.

As we are going towards to full digitalization of the banking services, banks also need to act with speed in the automation of banking services. It was noted that some of the services that customers would have wanted to be automated, they are still being done at the banking halls. The banks therefore need to work out on this shortfall in order to reach out to many customers that need these services. Lastly, some of the key informants highlighted that some of the challenges that they are facing are due to the issues to do with their third parties like the mobile network operators or internet service

providers and some of the challenges are even beyond their said counter parts. It is therefore high time that the government also takes a leading role in order to make sure that the infrastructure for this country is up to date in order to match the standards that customers are looking for.

5.3 Limitations

This study only explored the challenges and strategies that can be realised in the digital era; it only concentrated on banks' and customers' perspectives. But for the digitalisation drive to be achieved, other equally important players have a big role to play in the acceleration of digital banking growth in the country such as internet service providers, debit card issuers, software and hardware providers and others. As much as banks and customers are the critical players in the digitalisation drive, these players also have a role that they play. Since this research only focused on customers and banks, there is a need for further researchers to also explore the role that these third parties play in the acceleration of digital banking growth.

5.4 Areas for Further research

For future studies, it is recommended that all banks in the country take part in such research to have a better understanding of the issues that banks in the country face in the acceleration of digital banking. This will also help policymakers to make informed decisions on how best they can do in the drive towards a good user experience of online banking. The research was also cross sectional, it did not cover all the urban and district banks. In future researches, it is recommended that a similar research should be conducted in order to get views of all the banks both urban and district on how they view the direction of digital financial services in the country. Lastly, it emerged during the research that some of the challenges that the banks are facing are beyond them, these are issues to do with their third party service providers. It is therefore important that a similar research should also involve these third parties like the Mobile Network Operators, Natswitch and other players on what they are doing in the acceleration of digital financial services in the country.

REFERENCES

- Aburbeian, A. M., Owda, A. Y., & Owda, M. (2022). A technology acceptance model survey of the metaverse prospects. *AI*, *3*(2), 285–302. https://doi.org/10.3390/ai3020018
- Adosi, C. (2020). Qualitative data collection instruments: The most challenging and easiest to use. *Institute for Educational Planning and Administration*. https://www.researchgate.net/publication/344251614
- Al-Ajlouni, & Ahmed.(2018, April 12). Financial technology in banking industry: Challenges and opportunities. *e International Conference on Economics and Administrative Sciences*. http://dx.doi.org/10.2139/ssrn.3340363
- Ali, M., Khan, M. A., & Kalwar, M. A. (2021). Challenges for online banking in customers' perspective: A review. *International Journal of Business Education and Management Studies*, 5(1), 37–57. https://ijbems.com/doc/IJBEMS-167.pdf
- Almeida, F., Superior, I., Gaya, P., Queirós, A., & Faria, D. (2017). Strengths and Limitations of Qualitative and Quantitative Research Methods. *European Journal of Education stucdies*, 3(9), 370-374.
- Andrade, C., Gillen, M., Molina, J. A., & Wilmarth, M. J. (2022). The social and economic impact of covid-19 on family functioning and well-being: Where do we go from here? *Journal of Family and Economic Issues*, 43(2), 205–212. https://doi.org/10.1007/s10834-022-09848-x
- Aspers, P., & Corte, U. (2019). What is qualitative in qualitative research content courtesy of springer nature. https://doi.org/10.1007/s11133-019-9413-7%0AWhat
- Banke, N. K., & Yitayaw, M. K. (2022). Deposit mobilization and its determinants: Evidence from commercial banks in Ethiopia. *Future Business Journal*, 8(1). https://doi.org/10.1186/s43093-022-00144-6
- Borges, L., Perun, M., & Daradkeh, Y. (2020). Digital transformation and customers services: The banking revolution. *International Journal of Open Information Technologies*, 8(7), 124–128. https://cyberleninka.ru/article/n/digital-transformation-and-customers-services-the-banking-revolution

- Braun, V., & Clarke, V. (2006). Using thematic analysis in psychology. *Qualitative research in psychology*. 2, 77–101. https://doi/10.1191/1478088706qp063oa
- Bunker, G. (2020). Targeted cyber attacks: How to mitigate the increasing risk. *Network Security*, 2020(1), 17–19. https://doi.org/10.1016/S1353-4858(20)30010-6
- Chindudzi, G., Maradze, T., & Nyoni, T. (2020). The impact of digital banking services on the performance of commercial banks in Zimbabwe. *Journal of Management Information and Decision Sciences*, 6(6), 343–353. https://doi.org/10.7176/JMIDS.vol6.no6.19
- Crick, J. M. (2021). Qualitative research in marketing: What can academics do better? *Journal of Strategic Marketing*, 29(5), 390–429. https://doi.org/10.1080/0965254X.2020.1743738
- Cyril, U. S., & Cyril, U. S. (2019). *An overview of big data analytics in banking: Approaches , challenges and Issues.*. https://doi.org/10.33107/ubt-ic.2019.270
- Dasho, A., Meka, E., Sharko, G., & Baholli, I. (2016). Digital banking the wave of the future.

 Paper presented at the 7th International Conference organized by the department of statistics and applied informatics of the Faculty of Economy, University of Tirana. *International Conference Proceedings Digital*, 12(2), 1–6.

 https://www.researchgate.net/publication/315706647
- Devulapalli, S., & Oruganti, S. K. (2021). Challenges and opportunities of e-banking in India. *IOSR Journal of Business and Management*, 56–61. https://www.iosrjournals.org/iosr-jbm/papers/Conf.17037-2017/Volume-4/10.%2056-61.pdf
- Dickson, C., & Priyar, J. (2023). Digitisation of bank services and its impact on customer satisfaction and bank growth: A case study of NBS Bank plc. *International Journal of Humanities, Social Science and Management*, 3(2), 127-134.
- Fetu, A. (2019). Opportunity and challenges of electronic-banking system in commercial bank of Ethiopia (a case study on gurage zone). *Journal of Accounting Finance and Auditing Studies*, 5(2), 106–122. https://doi.org/10.32602/jafas.2019.23

- Festus, N., & Amaechi, J. (2019). Factors Influencing Banks 'Implementation and Consumers 'Acceptance of E- Factors Influencing Banks 'Implementation and Consumers 'Acceptance of E-Banking of Selected Commercial Banks in Calabar, Cross River State, Nigeria.

 October 2014. International Journal of Managerial Studies and Research, 2(9), 1-13.
- Fitzgerald, M., Kruschwitz, N., Bonnet, D., & Welch, M. (2013). *Embracing digital technology*. *a new strategic imperative*. embracing_digital_technology_a_new_strategic_imperative.pdf
- Hafeez-baig, A., & Chakraborty, S. (2016). Systems: Assuring reliability in qualitative studies: A health informatics assuring reliability in qualitative studies. *Proceedings of the 20th Pacific Asia Conference on Information*. http://www.pacis2016.org/abstract/Index.
- Kottner, J., Halfens, R., & Dassen, T. (2009). An interrater reliability study of the assessment of pressure ulcer risk using the Braden scale and the classification of pressure ulcers in a home care setting. *International Journal of Nursing Studies*, 46(10), 1307–1312. https://doi.org/10.1016/j.ijnurstu.2009.03.014
- Hunt M, All K, Burns B, Li K. T. (2021). Too much of a good thing: Who we follow, what we do, and how much time we spend on social media affects well-being. *J Soc Clin Psychol*, 40(1)46-68
- International Labour Organisation. (2021). Digitalization and the future of work in the financial services sector, Issues paper for the technical meeting on the impact of digitalization in the finance sector. file:///C:/Users/User/Downloads/wcms_824708%20(1).pdf
- International Monetary Fund. (2021). *The economic context of Botswana*. https://www.lloydsbanktrade.com/en/market-potential/botswana/economical-context
- Kala Kamdjoug, J. R., Ndassi Teutio, A. O., Tchakounte Tchouanga, U., & Gueyie, J.-P. (2022). use of e-banking and customer e-engagement in developing countries. *Theoretical Economics Letters*, *12*(05), 1378–1406. https://doi.org/10.4236/tel.2022.125076.
- Kaur, S. J., Ali, L., Hassan, M. K., & Al-Emran, M. (2021). Adoption of digital banking channels in an emerging economy: Exploring the role of in-branch efforts. *Journal of Financial Services Marketing*, 26(2), 107–121. https://doi.org/10.1057/s41264-020-00082-w
- King, B. (2018). Bank 4.0: Banking everywhere, never at a bank. Bank 4.0, 27. Cavendish Business.

- Majanga, B. B. (2016). The journey to financial inclusion in malawi- what does the future hold? *International Journal of Economics and Financial Research*, 2(9), 169–175.
- International Finance Corporation (2021). *Malawi economic monitor: From crisis response to a strong recovery*. World Bank
- National Planning Commission(2020). *Malawi's Vision 2063: An inclusively wealthy and self-reliant nation*. https://malawi.un.org/sites/default/files/2021-01/MW2063-%20Malawi%20Vision%202063%20Document.pdf
- National Statistical Office (2018). *Malawi Population and Housing Census*, 2018. www.nsomalawi.mw/index.php?option=com_content&view=.
- Manser Payne, E., Peltier, J. W., & Barger, V. A. (2018). Mobile banking and AI-enabled mobile banking: The differential effects of technological and non-technological factors on digital natives' perceptions and behavior. *Journal of Research in Interactive Marketing*, 12(3), 328–346. https://doi.org/10.1108/JRIM-07-2018-0087
- Ma, Q., & Liu, L. (2011). The technology acceptance model. Advanced topics in end user computing. https://doi.org/10.4018/9781591404743.ch006.ch000
- Mohd Arifin, S. R. (2018). Ethical considerations in qualitative study. *International Journal of Care Scholars*, *1*(2), 30–33. https://doi.org/10.31436/ijcs.v1i2.82
- Mokoena, S. M. (2020). *The Internet in Malawi: Past, present, and beyond.*https://malawi24.com/2020/05/22/the-internet-in-malawi-past-present-and-beyond/
- Mwiya, B., Chikumbi, F., Shikaputo, C., Kabala, E., Kaulung'ombe, B., & Siachinji, B. (2017). Examining factors influencing e-banking adoption: Evidence from bank customers in Zambia. *American Journal of Industrial and Business Management*, 7(6), 741–759. https://doi.org/10.4236/ajibm.2017.76053
- Nkuna, O., Lapukeni, A. F., Kaude, P., & Kabango, G. (2018). The role of commercial banks on financial inclusion in Malawi. *Open Journal of Business and Management*, 6(4), 813–832. https://doi.org/10.4236/ojbm.2018.64061

- Ofosu-Ampong, K. (2021). Determinants, barriers and strategies of digital transformation adoption in a developing country covid-19 era. *Journal of Digital Science*, *3*(2), 67–83. https://doi.org/10.33847/2686-8296.3.2_5
- Rabiu, I. D., Ladan, S., Usman, H. A., & Garba, M. (2019). Impact of e-banking on the operational efficiency of banks in Nigeria. *International Journal of Academic Research in Business and Social Sciences*, 9(2), 1–10. https://doi.org/10.6007/ijarbss/v9-i2/5527
- Rahman, M. M., & Rani Saha, A. (2018). Automated teller machine card fraud of financial organizations in Bangladesh. *Journal of Computer Science Applications and Information Technology*, *3*(1), 1–6. https://doi.org/10.15226/2474-9257/3/1/00126
- Rana, P., & Pandey, D. (2015). *Challenges and issues in e-banking services and operations in developing countries*. https://doi.org/10.4018/978-1-4666-9720-1.ch013
- Reserve Bank of Malawi, (2020). *Measures to mitigate the impact of COVID-19 from the banking industry in Malawi*. https://www.rbm.mw/Dynamic/SearchResult/
- Reserve Bank of Malawi. (2022). Monthly Economic review. Reserve Bank of Malawi.
- Revathi, P. (2019). Digital banking challenges and opportunities in India. *EPRA International Journal of Economic and Business Review*, 7 (12), 20–23. https://doi.org/10.36713/epra2985.
- Selvam, P., Premnath, S., Prasanth, S., Sudhamathi, D. S., & Scholar, R. (2018). Challenges faced by banking sector on digital innovation. *Journal of Emerging Technologies and Innovative Research*, 5 (10), 964-966.
- Severino, M., Tonderai, N., & Life, S. (2015). Impact of mobile banking on traditional banking. *International Journal of Economics, Commerce and Management*, *3*(1), 1–13.
- Sharma, B., & Dubey, A. (2022). Digital banking: A need of time. *International Journal of Advance and Applied Research*, 9 (3), 504-519

 https://www.researchgate.net/publication/360809100

- Shcherbatykh, D., Shpileva, V., Riabokin, M., Zham, O., & Zalizniuk, V. (2021). Impact of digitalization on the banking system transformation. *International Journal of Computer Science and Network Security*, 21(12), 513-520. https://doi.org/10.22937/IJCSNS.2021.21.12.71
- Skinner, C. (2014). *Digital bank: Strategies to launch or become a digital bank*. Cavendish International.
- Taiwo, J.N., & Agwu, M.E., (2017). The Role of E-Banking on operational efficiency of Banks in Nigeria. *Basic Research Journal of Business Management and Accounts*, 6(1), 1-10
- Taher, A., Sheikh & Kama. (2018). Use of big data in financial sector of Bangladesh: A review. 22nd Biennial Conference of the International Telecommunications Society: Beyond the Boundaries: Challenges for Business, Policy and Society. https://hdl.handle.net/10419/190348.

APPENDICES

APPENDIX 1: INTERVIEW GUIDE-BANKERS PERSPESTIVE

A guide for in-depth interview	WS	
E-banking opportunities and	challenges- Banker's p	perspective
Name of interview		
Date of interview		
Interviewee code		
Introduction		
	him/herself, describe	terviewee for their interest to participate in the main reasons for the interview, and explain the
Consent		
Tinyade Kayira, a Master of I and Applied Sciences (MUBA in the adoption of digital bank you face that encourage custe also like to know how support online banking and the opport to know what the bank is doin than the traditional banking. to audio-record the interview ideas. Since I will audio recowe do not miss any of your confidential, meaning, no narrecordings will be kept at a scare no right/wrong answers;	Business Administration (AS). I am interested to king financial services omers to visit the physicive is the bank to the fitunities that the banks g in order to migrate in The interview should by so that we can accurate your voice, please or comments. Please of the place, and destroyed we just want to hear also that you do not a questions about what the to participate in this services.	
Interviewee (signature)		Date

Questions guided by topic

1. Demographic information

I would like to start by asking you a few questions concerning the demographic information of your bank (The interviewer will then fill the responses in the table below)

Age range (years)	□18-30 □31-40	☐ 41-50 ☐51 & above
Gender	☐ Male	Female
Marital Status	☐ Single	☐ Married
	☐ Divorced/Separated	□Widowed
Education	☐ Diploma	☐ Bachelors
	☐ Masters	
Employment experience	□ 1-5	☐ 6-10
(years)	□ 11-15	☐16 & above

2. Performance of digital banking

I am now going to ask you some questions on what has been the trend of digital banking over the years. Tell me everything you know on how digital banking has been performing over the years? (Note to interviewer: after giving time to the interviewee to respond, probe specific planned questions he/she may not have addressed)

- a. How have the customers welcomed e-banking?
- b. Have you made significant progress over the years?
- c. What have been the developments since the launch of digital banking??
- d. What has been the feedback so far from the concerned stakeholders?
- e. Are you where you had intended to be now considering the number of years that digital banking has been existing at this bank?

3. E-banking opportunities

I am now going to ask you about views, opinions and suggestions considering your experience on this job on how best you think banks can take opportunities for digital banking growth in the country. What opportunities can you foresee that can accelerate the growth of digital banking? (Note to interviewer: after giving time to the interviewee to respond, probe specific planned questions he/she may not have addressed)

- a. As a bank, have you considered any collaborations with other service providers in order to enhance your service delivery?
- b. What opportunities does e-banking offer to your bank to expand your customer base?
- c. How can banks turn the security challenges associated with e-banking into opportunities?
- d. How has e-banking enhanced operational efficiency?
- e. How can e-banking improve financial inclusion since the RBM is advocating for a cashless society?

4. E-banking challenges

I now want you to talk about the challenges of e-banking services that you face as the bank and the challenges that are also frequently reported by customers. What would make customers not to use e-banking service? (Note to interviewer: after giving time to the interviewee to respond, probe specific planned questions he/she may not have addressed)

- a. What are the challenges that you are facing as a division?
- b. What are the challenges that are frequently reported by your customers?
- c. Do you get constant feedback from the customers on the performance of digital banking platforms? Are the challenges timely addressed?
- d. In your opinion, what do you think still makes customers to visit traditional banks to digital banks?

5. E-banking strategies

I am now going to ask you about your views, opinions and suggestions on how best you think digital banking can progress in the country. What strategies do you think as the bank should put in place in order to stay competitive in the digital age? (Note to interviewer: after giving time to the interviewee to respond, probe specific planned questions he/she may not have addressed)

- a. What role do you think your customers have that can help to accelerate the growth of digital banking?
- b. Since the world globally is going digital, what plans have you put in place that will assist you to compete globally?
- c. What effective forms of communication do you use in order to get constant feedback from your customers? Are the issues timely addressed?
- d. Do you use any incentives in order to migrate more customers to online banking?
- e. What security measures have you put in place in order to make sure that your customers are protected?

Is there anything you would like to add?

Thank you for your participation

APPENDIX 2: INTERVIEW GUIDE-CUSTOMER PERSPECTIVE

A guide for in-depth intervi	ews	
E-banking opportunities and	d challenges-Custon	mers' perspective
N		
Name of interview		
Date of interview		
Interviewee code		
Introduction		
	e him/herself, descr	he interviewee for their interest to participate in the ribe main reasons for the interview, and explain the
Consent		
Tinyade Kayira, a Master of and Applied Sciences (MUE in the adoption of digital bayou face that encourage you like to know your level of us your journey of online bank. The interview should not take so that we can accurately confine your voice, please make su comments. Please be assured be attached to the interview transcription is done. Reme experiences, so feel free to exto be forced to talk about an just said?	f Business Administ BAS). I am interested the physical services to visit the physical and also how you are so and that your responsives and the recording ember, there are not apture your feeling anything you do not	trerest to participate in this interview. I am Anna tration student at the Malawi University of Business and to hear about the challenges that you are facing vices. I would like to know the recurring issues that cal bank rather than the online bank. I would also be online banking and how supportive is the bank in you envision the future of e-banking in the country. I minutes. I would like to audio-record the interview ances, thoughts and ideas. Since I will audio record udible enough so that we do not miss any of your ses will be kept confidential, meaning, no names will as will be kept at a safe place, and destroyed once to right/wrong answers; we just want to hear your sand thoughts. Remember also that you do not have want to. Are there any questions about what I have
Please sign below if you ago	ree to participate in	n this interview
Interviewee (signature)		Date

Questions guided by topic

1. Demographic information

I would like to start by asking you a few questions concerning your demographic information (The interviewer will then fill the responses in the table below)

Age range (years)	□18-30 □31-40	☐ 41-50 ☐ 51 & above
Gender	☐ Male	Female
Marital Status	☐ Single	☐ Married
	☐ Divorced/Separated	□Widowed
Education	☐ Primary	☐ Secondary
	☐ Tertiary	
Employment Status	☐ Unemployed	☐ Self-employed
	☐ Employed	☐ Still at school
How long have you held an		
account with this bank?		

2. Knowledge of digital banking

I am now going to ask you some questions about your knowledge of e-banking services. **Tell me everything that you know about e-banking?** (Note to interviewer: after giving time to the interviewee to respond, probe specific planned questions he/she may not have addressed)

- a. What are the types of e-banking services that you know?
- b. What security measures are you aware of when using e-banking?
- c. How do you stay updated to the new changes in e-banking?
- d. Do you know any other means of sending money other than using cash?
- e. Do you know other e-banking services like bill payments, tuition fee payments, tax payment, masm payment, DSTV payment and loan application?

3. Customer's e-banking services use

I am now going to ask you about your usage of e-banking services. **How is your usage of online banking so far like?** (Note to interviewer: after giving time to the interviewee to respond, probe specific planned questions he/she may not have addressed)

- a. How often do you use e-banking services?
- b. What type of transactions do you usually perform on e-banking?
- c. What motivates your choice of e-banking service over the other?
- d. How satisfied are you with the performance of online banking so far?
- e. What would make you use e-banking over the traditional banking?

4. E-banking challenges

I now want you to talk about your challenges of e-banking services that you face. What are the specific challenges that you are facing with e-banking? (Note to interviewer: after giving time to the interviewee to respond, probe specific planned questions he/she may not have addressed)

- a. Are e-banking platforms user-friendly to you?
- b. Have you reported any of the challenges that you face?
- c. Is the bank able to address these challenges?
- d. Are security issues, technical issues, lack of trust, connectivity problems, not hindering your e-banking usage?
- e. Is the bank available to take you through on the use of e-banking services?

a. The future of digital banking

I am now going to ask about your perceptions on what you think the banks in the country should do right in order to accelerate the digital banking growth. **How do you envision the future of e-banking?** (Note to interviewer: after giving time to the interviewee to respond, probe specific planned questions he/she may not have addressed)

- a. What new features or improvements would you like to see?
- b. What features do you find most useful in your current e-banking service that you would love the bank to continue improving?
- c. What current features of e-banking services you would want the bank to remove altogether?
- d. What measures would you like to see to enhance the security of online banking?
- e. What are your thoughts on digital/mobile wallets on the future of e-banking?

Is there anything you would like to add?

Thank you for your participation